Major Shareholders' Trust and Market Risk: Substituting Weak Institutions with Trust

Omneya Abdelsalam^{1,*}, Antonios Chantziaras^{1,2}, Jonathan A. Batten³, Ahmet Faruk Aysan⁴

Keywords: Trust, controlling shareholder, market risk, volatility, return stability.

JEL Classifications: G12, G32, G41.

Declarations of interest: none.

<u>Funding:</u> this research received funding from the El Shaarani Research Centre for Ethical Finance, Accountability & Governance at Durham University.

¹ Durham University Business School, Durham University, Mill Hill Lane, Durham, DH1 3LB, United Kingdom. ² School of Humanities, Social Sciences and Economics, International Hellenic University, 14th klm Thessaloniki-Moudania, Thessaloniki, 57101, Greece.

³ School of Economics, Finance and Firmingm Universiti Utara Malaysia, Sintok 0610, Kedah, Malaysia.

⁴ School of Management and Administrative Science, Istanbul Sehir University, Turgut Özal Bulvarı, No: 21, Dragos, Kartal/Istanbul, 34865, Turkey.

^{*} Abdelsalam (corresponding author), o.h.abdelsalam@durham.ac.uk; Chantziaras (^b https://orcid.org/0000-0002-1309-6967), antonios.chantziaras@durham.ac.uk; Batten, jabatten@gmail.com; Aysan, ahmetaysan@sehir.edu.tr.

Major Shareholders' Trust and Market Risk: Substituting Weak Institutions with Trust

Abstract

This study examines the impact of foreign controlling shareholder trust on firm market risk using the two measures of total and idiosyncratic risk. An extensive global sample of 12,496 firm-year observations from 43 countries is employed. The results show that firms controlled by foreign trusting shareholders display lower levels of risk in both market measures. Trust appears more important for firms based in countries with a less favourable institutional environment, whereby it varies with the investment horizon of foreign controlling shareholders. The results are robust after controlling for cultural measures, endogeneity, selection bias and alternative model specifications.

Keywords: Trust, controlling shareholder, market risk, volatility, return stability.

JEL Classification: G12, G32, G41.

1. Introduction

The influence of trust and social capital on individual behaviour has been extensively studied in the economics literature (see, e.g., Glaeser et al., 2000). In addition, there is a sizeable law and finance literature that focuses on corporate behaviour in diverse institutional settings (see, e.g., La Porta et al., 1997; Laeven and Levine, 2008). These studies show that common and civil law countries that uphold the rule of law through strong institutions and property rights allow individuals to dispense with costly monitoring. Despite the many studies in this area, little is understood about the impact of societal trust on the corporate risk profile. A key contribution of this study is that we take a global perspective and explore the impact of foreign controlling shareholders' trust on the market risk of firms. Our theoretical framework predicts that large foreign shareholders originating from high-trust environments reduce the market risk of firms (i.e., lower return volatility).

The monitoring of controlling shareholders is considered an important corporate governance tool that reduces agency costs (Jensen and Meckling, 1976; Taub, 2018). Corporate strategic decisions are often shaped by large controlling shareholders as they decide how management is to be monitored and compensated (Zou and Adams, 2008). Recent studies also provide evidence that a company's performance and risk profile are affected by the presence of large controlling shareholders, who can leverage their positions using convoluted systems of voting rights (Faccio et al., 2011; Mishra, 2011; García-Kuhnert et al., 2015; Taub, 2018).

The presence of foreign investors has been shown to reduce stock price volatility (Umutlu et al., 2010; Li et al., 2011; Vo, 2015). Stock market volatility is commonly analysed as an outcome of corporate decisions. High disclosure rankings, for example, are directly associated with lower subsequent stock return volatility (Bushee and Noe, 2000). Additionally, El Ghoul et al. (2019) find that firms located in countries with a high level of trust are more

likely to have a zero-leverage policy indicating that a country's culture affects corporate financial outcomes by influencing the actions of firm stakeholders. Consequently, the analysis of the influence of trust, and its association with the presence of large controlling shareholders, on the risk of firm market returns remains an important empirical task and is the primary focus of this paper.

In this study the trust associated with the presence of foreign shareholders is measured by the level of societal trust in the countries where the major shareholders are domiciled. Specifially, we measure a country's level of societal trust based on data from the World Values Survey, which is a proxy widely employed in recent studies (e.g., Georgarakos and Pasini, 2011; Kanagaretnam et al., 2018; Kanagaretnam et al., 2019). Consistent with the literature, we employ two measures of risk, namely total risk (the standard deviation of each firm's weekly stock returns) and idiosyncratic risk (the standard deviation of the residuals from the Fama and French (1993) three-factor model).

Our study builds on an important strand of the literature which has considered the effect of trust on economic decisions since trust underlies all commercial transactions and economic exchanges (Arrow, 1972; Williamson, 1993). Kanagaretnam et al. (2017) summarize this literature and show that higher levels of trust have positive macroeconomic (such as facilitating economic growth and international trade) and financial market (such as improving corporate financing and mergers and acquisitions) benefits as well as decreasing individual firm risk.

The study employs a global sample from 43 countries and 12,496 firm-year observations for the 2000-2016 period. The key finding is the negative effect of foreign controlling shareholders' trust on firm market risk, after considering for firm and country-level characteristics and controlling for industry and year effects. The evidence shows that firms controlled by large foreign shareholders originating from high-trust environments display

lower levels of risk using both market measures. We also find that trust is more important for firms located in countries with a less favourable institutional environment as well as for firms controlled by foreign shareholders with a long-term horizon. These results are robust after controlling for various cultural measures. We also consider the endogeneity of societal trust, which has a bidirectional version of causality with institutional quality (Knack, 2002; Paxton, 2002). To address the potential endogeneity concern, we apply a two-stage least squares analysis (2SLS) to ensure that unobservable determinants of trust do not influence firm market risk. Our findings are also robust to alternative model specifications and measures of firm risk as well as for selection bias and alternative sample constructs.

Our paper makes several important contributions to the law and finance and social capital literature. Firstly, we can provide empirical insights into how trust affects corporate market risk. Secondly, this study takes an international perspective, thereby adding alternative and more complex legal and property rights perspectives to various existing single country studies, including those in the United States (see, e.g., Hasan and Habib, 2019). Thirdly, we enhance the understanding of the effect of trust on market risk. Although existing studies investigate the relationship between trust and risk, they either disregard the financial sector or use accounting-based measures of firm risk (see, e.g., Kanagaretnam et al., 2017; Kanagaretnam et al., 2019). Given that risk, measured by total and idiosyncratic risk in our setting, has a negative impact on shareholder value, it is important to understand its determinants. This is especially important for idiosyncratic risk, which constitutes the largest component of individual stock returns (Goyal and Santa-Clara, 2003; Lui et al., 2007). Fourth, we contribute to a new strand of the literature that investigates the effects of trust on corporate decision-making (see, e.g., Pevzner et al., 2015; Bottazzi et al., 2016; Cingano and Pinotti, 2016; Dudley and Zhang, 2016; Qian et al., 2018; Hasan and Habib, 2019).

Fifth, we extend the understanding of the effect of the shareholder horizon on firm outcomes (Bushee and Goodman, 2007; Chen et al., 2007; Harford et al., 2018; Ding et al., 2020; Döring et al., in press) by indicating that long-term oriented foreign shareholders from high-trust countries reduce firm market risk. Sixth, by using a global sample, the study can compare various regions and different institutional environments, thereby adding to the already extensive law and finance literature. This study is, therefore, one of the first to consider the impact of the *trust* of large foreign shareholders on organisations' outcomes. Seventh, we propose that the geographic location where the controlling shareholders are based, and its degree of societal trust will significantly influence the level of firm risk. Thus the paper adds to the relevant literature on the supplementary role of informal institutions (North, 1994; Pevzner et al., 2015; Qian et al., 2018). Our empirical investigation indicates that trust plays a more important role in regions where formal institutions are less effective, and thus, acts as a substitute for the presence of formal institutions. Overall, we claim that higher levels of trust in the country of origin of foreign shareholders instigate them to hold corporate shares for longer (inaction) and, consequently, this lowers the market risk of firms.

The remainder of our paper is organized as follows. In Section 2, we review the related literature. Section 3 describes our research design and sample, while Section 4 describes our empirical results. Section 5 reports the extensions of our empirical investigation, and Section 6 presents the sensitivity testing and robustness of our results. Finally, Section 7 concludes the article.

2. Theory and Hypothesis Development

Trust, as a social and moral concept, and its link to individual and corporate risk-taking have attracted academic interest in economics and finance since it underlies all commercial transactions and economic exchanges (Arrow, 1972; Williamson, 1993). The concepts of trust

and risk are closely related: more trust is needed in risky situations and trusting someone or something (such as an organization) in itself involves taking a risk (Sabel, 1993). Thus, the nexus between risk and trust has also gained increasing acceptance among a number of social theorists (see, e.g., Gambetta, 1988; Mayer et al., 1995).

2.1 Large Shareholders and Control

It has been empirically established that major shareholders can act as monitors to alleviate agency problems. They do so with the aim of preventing actions that are in conflict with their own interests (DeMarzo and Urošević, 2006; Cronqvist and Fahlenbrach, 2009) through direct action, negotiation with management, and proxy fights (DeMarzo and Urošević, 2006). Managers who repeatedly act against the desires of controlling shareholders are likely to be replaced (Andres, 2008). Thus, it is expected that better monitoring reduces firm risk and uncertainty, increases output and positively influences firm value (DeMarzo and Urošević, 2006).¹

There are, however, possible disadvantages to concentrated ownership. Firstly, agency costs will increase if large shareholders seek their own interests. This negatively impacts other stakeholders, such as minority shareholders and employees (Gürsoy and Kürşat, 2002; Andres, 2008). As a result, large shareholders could cause a deterioration in firm value (Edmans, 2014). Secondly, the firm's investment policy could be affected as concentrated ownership could indicate a lack of diversification with high exposure to systematic risk. Overall, the empirical evidence demonstrates that large shareholders can affect firm investment and financial policies

¹ When ownership is focused on a few individuals, agency costs are reduced as the monitoring of managers is easier and more efficient. Based on theoretical modelling, concentrated share ownership allows the largest shareholder to monitor managerial efforts more closely and modify the compensation contract accordingly (La Porta et al., 1999; Claessens et al., 2002; Faccio and Lang, 2002). Thus, when large shareholders control companies, their influence on corporate behaviour is expected to be higher (Taub, 2018).

(Goergen and Renneboog, 2001; Gürsoy and Kürşat, 2002; Cronqvist and Fahlenbrach, 2009) and contribute directly to higher risk (Gürsoy and Kürşat, 2002).

More recent studies also report that the risk profile and performance of companies are significantly impacted by large controlling shareholders' investment portfolios (Faccio et al., 2011; Mishra, 2011; García-Kuhnert et al., 2015). Such studies show that firms controlled by large diversified shareholders participate in riskier investments (Faccio et al., 2011). When similar tests were conducted on the banking sector, the results were found to be consistent (García-Kuhnert et al., 2015). Comparably, Paligorova (2010) found that corporate risk-taking is positively affected by shareholders with large equity stakes in several companies. This is further supported by Mishra (2011), who concludes that only large and diversified shareholders engage in riskier investment strategies.

Bushee (1998) shows that institutional investors do not reduce R&D expenditure, implying that they are sufficiently sophisticated to play a monitoring role in reducing myopic behaviour. This conclusion is confirmed in a series of papers by Bushee (2001, 2004). The literature also informs that shareholder monitoring intensity is conditional on their investment horizon. For example, dedicated (long-term oriented) investors are more active in monitoring since they have long-term interests in firms and often invest large stakes (e.g., Bushee, 1998; Chen et al., 2007).

On the other hand, transient (short-term oriented) investors are associated with frequent trading, and this short-term orientation, without much monitoring, encourages managers to engage in myopic behaviour (e.g., Bushee and Goodman, 2007; Brockman and Yan, 2009). Cumming et al. (2020) explore the effect of public-to-private buyout transactions on investments in innovation. Buyouts are associated with a significant reduction in patents and patent citations. However, the negative effect of buyouts is established only for institutional

buyouts, indicating that only institutional buyouts prevent target firms from adopting long-term investments. Attig et al. (2012) investigate the relevance of institutional investors' investment horizon to corporate governance and show that institutional investors with longer investment horizons engage in more effective monitoring. This benefits the firm since Attig et al. (2013) later demonstrate that firms with institutional investors with longer investment horizons tend to have a lower cost of equity due to improved monitoring and information quality. Hence, the investment horizons of foreign investors are critical and are examined in the empirical section of this paper.

Existing theoretical and empirical evidence shows that corporate behaviour may also be affected by the individual characteristics of large controlling shareholders. For example, the presence of foreign large controlling shareholders may improve the market-wide accuracy of public information (Umutlu et al., 2010) and enhance corporate governance and monitoring (Li et al., 2011). Additionally, the presence of foreign investors may improve the quality of risk controls and reduce the risk exposures of firms, especially those in emerging markets (Mitton, 2006; Ferreira and Matos, 2008; Umutlu et al., 2010). The prior literature highlights the "stabilizing" role of large foreign shareholders on emerging stock markets (Wang, 2007; Li et al., 2011) as their presence has a negative association with stock return volatility. Foreign investors also foster long-term investment in tangible, intangible and human capital (Bena et al., 2017). Moreover, the external social factors that shape shareholders' individual characteristics are also likely to have important implications for corporate performance. One important feature of this study is that we examine the impact of foreign shareholder trust on firm market returns.

2.2 Trust and Risk

The concept of trust is subjective and incorporates the individually assessed probability that the trustee will perform as expected (Gambetta, 1988). The subjective view of trust affects behavioural trust, which is the action of trust. Thus, subjective trust associates, first, with risk perception, which is considered as the probability that the trustee will not fully commit to a relationship and will act opportunistically, and second, with the probability and consequences of not achieving the goals in a relationship (Das and Teng, 2004). These two dimensions of risk affect the likelihood of risk-taking. We propose that there is an inverse relationship between perceived risk and subjective trust. Under this perspective, subjective trust and perceived risk can be considered as complimentary. At the one extreme, if there is no risk, there is no need for subjective trust. However, as the risk perception increases, the need for subjective trust increases as well.

Trust antecedents (i.e., personality, institutional and situational characteristics) directly influence the level of subjective trust, or the propensity to take the risk with the trustee. The prior literature suggests that values and networks in a society facilitate socially responsible behaviour and enhance societal trust (La Porta et al., 1997; Buonanno et al., 2009). Formalizing institutional rules (restraining arbitrary behaviour by government leaders) and reducing social polarization have a positive impact on societal cooperation and trust (Knack and Keefer, 1997). The enhanced stability of democratic institutions is an important factor for the development of societal trust (Inglehart, 1999). Indicatively, areas with high trust tend to have a stronger rule of law (La Porta et al., 1997) and less criminal activity (Buonanno et al., 2009). In summary, behavioural trust inevitably invites risk-taking.

On the other hand, Chircop et al. (2020) show that there is a positive relationship between religiosity and risk aversion, with venture capitalists located in more religious counties making less risky investments. They prefer to be involved in staging and syndication and to invest in the later and expansion stages of portfolio companies.

Williamson (2000) provides a four-level social analysis to conceptualize various layers of institutions. Brockman et al. (in press) further elaborate on these institutional layers while testing their impact on security-level contract design in U.S. debt markets. At Level 1, which includes social trust, institutions are highly persistent over time and inherent to the country's underlying culture and norms. Formal political, judicial and economic institutions, such as investor protection and the political institutions of a country, are considered at Level 2. Level 3 concerns the individual and firm-level governance structures. The economic outcomes of the first three levels of institutions are revealed with transaction-level quantities and prices and the allocation of resources and thereby form the fourth level of economic institutions. In our paper, the aggregate and idiosyncratic risk measures are the Level 4 economic variables. Similarly, El Ghoul et al. (2017) provide insights into non-market mechanisms, such as CSR, through which firms can compensate for institutional voids. In the same vein, Schoorman et al. (2007) discuss substitution effects between control and trust in organizations.

Social trust construct is applied to other economic outcomes (Level 4) in explaining cross-country differences. Guiso et al. (2004) investigate the relationship between financial development and social trust. Similarly, Guiso et al. (2006) analyse the link between social trust and national savings, entrepreneurship, income redistribution, and finally stock market participation (Guiso et al., 2008). Bottazzi et al. (2016) concentrate on the effects of social trust on the behaviours of venture capitalists, while Pevzner et al. (2015) explore the effects of trust on stock market reactions to earnings announcements. The availability of trade credit during financial crises in the context of social trust is analysed by Levine et al. (2018). Finally, Brockman et al. (in press) analyse the role of social trust in international contracting.

2.3 Linking Trust with Market Risk

Risk reflects uncertainty about outcomes or events. At the firm level, the degree of risk incurred by a firm is crucial for investors, in particular, the number of financial performance fluctuations over time, such as in the share price (market risk) or internal accounting returns (accounting risk). Overall, firm risk is important because it can impair forecasting and planning activities, while it may indicate higher variability in organizational returns and/or a higher propensity for corporate decline and mortality (Baird and Thomas, 1985; Fiegenbaum and Thomas, 1988; Miller and Bromiley, 1990). Specifically for financial institutions, excess institutional and individual risk-taking may be one of the main causes of financial crises and volatility in financial markets (Jordà et al., 2013).

What emerges from the above discussion is that market risk is an important consideration for investors. According to traditional asset pricing theory, investors can diversify market risk through portfolio diversification. However, previous research suggests that investors do not often hold perfectly diversified portfolios for several reasons, such as the high transaction and search costs (Ang et al., 2006); a preference for exposure to a single firm; wealth constraints (Xu and Malkiel, 2003); preferences for stocks with certain characteristics (i.e., higher volatility, higher turnover etc.) or *erroneous* diversification strategies (Goetzmann and Kumar, 2008); and limited investor sophistication and investor-specific attributes (Dorn and Huberman, 2005).

Considering these reasons, it becomes obvious that investors cannot hold "fully" diversified portfolios. Since under-diversification exposes investors to higher risk, understanding the determinants of firm market risk is quite important for maximizing investors' wealth. Firm idiosyncratic risk arising from factors such as a deterioration in financial reporting quality and cash flow uncertainty (Irvine and Pontiff, 2009; Rajgopal and Venkatachalam,

2011) represents an important cost for investors and can negatively affect investors' wealth (Pontiff, 2006). More importantly, firm idiosyncratic risk results primarily from a firm's actions and is independent of the common market movement, and thus constitutes a factor that increases investors' perception of risk.

Trust is also directly linked with the level of market risk. Trust brings with it consistency over time and persistence in investment and thereby lower volatility (Mayer et al., 1995; Hasan and Habib, 2019). When shareholders have trust, they do not change their minds or their investment strategies often and abruptly (Schoorman et al., 2007). Trust in others can act as a substitute for costly monitoring (e.g., Knack and Keefer, 1997; Zak and Knack, 2001) and thus enhance economic performance, because economic agents in high-trust environments devote more time to production rather than costly monitoring (Lesmeister et al., 2018). Higher levels of trust also enhance stock market participation (Balloch et al., 2015) and increase stock ownership (Georgarakos and Pasini, 2011). Bottazzi et al. (2016) show a positive relationship between trust and investment, and their theory predicts that earlier stage investments require higher trust.

A lack of trust leads to a delay in investment decisions, or once the investment is carried out, it leads to frequent changes in attitudes towards the investment while leading to increased volatility in stock prices. Georgarakos and Pasini (2011) highlight that mistrust lowers the expected return from investments as investors link a lack of trust with an increased possibility that a contract will not be respected by the counterpart. A delay in investment decisions resulting from a lack of trust also implies that arbitrage opportunities in the market are not immediately materialized due to low participation in the market, thereby making the market even more volatile.

2.4 Theoretical Relationship and Hypothesis

According to Luhmann's (1979) theory of social systems, social trust provides a mechanism for reducing the internal complexity of social interaction and enables actors to establish mutual expectations of future behaviour (Dyer and Chu, 2003). We propose that the key features of an institutional environment, (i.e., investor protection, economic development and stock market development) and the geographical location of controlling shareholders in conjunction with the characteristics of the trustee (i.e., controlled by trusting foreign investors with a long-term horizon) will have a direct relationship on the propensity to take risk with the trustee and will hence lead to less volatility in the latter's share prices. For example, Allen (2005) annotates that China, a country with weak institutions, has experienced strong economic growth due to strong informal institutions, such as trust, acting as a substitute for good corporate governance and strong laws.

This proposition suggests that trust is important in cases where the trustor cannot take sufficient control of the situation (Nooteboom, 1996). Following the rationale of Guiso et al. (2008) and Zak and Knack (2001), we anticipate that higher levels of trust will increase shareholders' participation in the stock market and will instigate managers to act on behalf of shareholders' interests (due to the abovementioned cost of cheating). As a result, shareholders will be willing to hold corporate shares for longer, which in turn will reduce the market risk of firms (i.e., lower return volatility). The higher level of trust will encourage controlling shareholders to *place trust* in the actions of the trustee (in our case, the firm) (Michalos, 1990) and take more risk (which is measured by firm market risk).

Higher trusting shareholders also have a long-term horizon for the firm (Brockman et al., in press) and make more long-term investments (Bottazzi et al., 2016), while El Ghoul et al. (2019) highlight the importance of a country's culture in affecting corporate financial

outcomes. Our paper focuses on the social trust levels of foreign shareholders, who conduct more rigorous internal procedures when investing in other countries as compared to in their home countries. Hence, due to these transaction costs, once foreign shareholders decide to invest in a foreign country as blockholders, they prefer to stay longer in the host country and foster long-term investment in tangible, intangible and human capital (Bena et al., 2017). In addition, higher trusting foreign shareholders invest even more in long-term projects (Bottazzi et al., 2016). This long-term perspective reduces the return volatilities of the firm.

When foreign shareholders invest in a company, the company is subject to different managerial standards and norms. This is due to foreign investors being "less encumbered by ties with corporate insiders" (Bena et al., 2017, p. 123), a fact that curbs managerial entrenchment. Foreign-controlled firms often operate differently because of the international horizon they adopt. These social norms make them less volatile on the supply side, and the perception of this pattern on the demand side also lowers their risk or return volatility. The alternate view would be that despite societal trust contributing to the reduction of risk, it might not be enough to eliminate relational and performance risks. Under this perspective, investors will still be vulnerable to opportunistic behaviours and exposed to potential losses from financial performance fluctuations, which would be detrimental to their wealth. Sound examples include the collapse of Lehman Brothers and the rescue of AIG by the U.S. government (in September 2008), which yielded severe losses for investors. However, the empirical evidence indicates that the level of trust in the financial sector was "indeed very low during the last months of 2008" (Sapienza and Zingales, 2012, p. 130), strengthening the argument of low investor participation in stock markets when levels of trust are low (Guiso et al., 2004, 2008). Based on the discussion above, we formulate the following hypothesis:

H: We propose that firms controlled by foreign shareholders that are domiciled in more trusting environments are more likely to have lower risk in stock returns after controlling for differences in the institutional environment.

3. Research Design

3.1 Measuring Firm Market Risk

We estimate firm market risk using the following two measures: total risk and idiosyncratic risk (Sila et al., 2016; Abdoh and Varela, 2017). Total risk (TRISK) is calculated as the yearly standard deviation of a firm's weekly stock returns. We follow Ang et al. (2006) and estimate idiosyncratic risk using the Fama and French (1993) three-factor model, as shown below:

$$RETURN_t = \alpha + \beta_1 MARKET_RETURN_t + \beta_2 SMB_t + \beta_3 HML_t + \varepsilon_t$$
⁽¹⁾

RETURN and MARKET_RETURN are the weekly stock and index returns, respectively. SMB is the difference between the excess return on a portfolio of small versus large capitalization stocks. HML is the difference between the excess return on a portfolio of high versus low book-to-market stocks, ε is the error term measuring firm-specific risk, and the subscript *t* references time *t*. Following Amihud (2002), stocks must have at least 40 non-missing trading months. We estimate model (1) for each firm and each year and calculate idiosyncratic risk as to the standard deviation of the residuals (IRISK). Following Sila et al. (2016), we annualize both measures of risk and multiply them by the square root of 250.

3.2 Measuring Foreign Shareholders' Trust

The measurement of shareholder country trust begins by identifying the large shareholders in each company. Large shareholders are defined as those who hold the largest stake of voting rights, under the condition that this stake is at least 5%. There is no unanimous

theoretical ground for defining blockholders, but normally, they are defined as shareholders that hold at least 10% (Bodnaruk et al., 2008; Laeven and Levine, 2008; Mishra, 2011) or at least 5% (Li et al., 2006; Cronqvist and Fahlenbrach, 2009) of the equity ownership of the firm. The cut-off of 5% is conventionally used in the literature as it represents the minimum significant threshold of votes. Also, most countries mandate the disclosure of 5% ownership stakes.

After identifying the largest controlling shareholder for all companies with available ownership data, we match the shareholders' countries with societal trust data². This approach is consistent with a number of other studies (see, e.g., Guiso et al., 2008; Georgarakos and Pasini, 2011; Pevzner et al., 2015; Cingano and Pinotti, 2016; Dudley and Zhang, 2016) and is based on the following question from the World Values Survey (WVS): "Generally speaking, would you say that most people can be trusted or that you need to be very careful in dealing with people?". This trust measure represents the average fraction of respondents that answer "Most people can be trusted" in each country over three surveys conducted in 1999-2004, 2005-2008, and 2010-2014³. Hence, shareholder country trust (SHTRUST) represents the level of societal trust where the shareholder is based, which is expected to have a strong influence on the characteristics of the shareholders.

Throughout our study, we concentrate on large foreign shareholders, whereby a foreign shareholder is defined as a citizen of another country, a business entity registered (or headquartered) in another country, or an unlisted majority-owned subsidiary of a foreign

 $^{^{2}}$ We obtain data for each firm's ownership structure on a yearly basis (i.e., at the end of each calendar year – 31st of December), through Thomson Reuters Eikon database. Therefore, our data capture potential changes in the ownership structure of each firm across the entire sample period (i.e., 2000-2016). For each shareholder's country of origin, we rely upon Eikon's identification.

³ The seventh wave of the WVS commenced in January 2017 and lasted until December 2019 (see http://www.worldvaluessurvey.org/wvs.jsp, Accessed 10 February, 2020). Since WVS waves are planned every five years, we consider the values of 2015 and 2016 to be equal to the sixth wave (2010-2014). Our inferences remain unchanged if we remove 2015 and 2016 from the analysis or when we interpolate SHTRUST values between two adjacent waves.

company (similar to Li et al., 2011). We focus on foreign shareholders since earlier studies demonstrate that foreign investors help to reduce risks in firms that they invest in and hence, reduce stock price volatility (Umutlu et al., 2010; Li et al., 2011). This phenomenon has been documented by prior studies (e.g., Vietnam (Vo, 2015), Indonesia and Thailand (Wang, 2007), and in 31 emerging markets (Li et al., 2011)). The entry of foreign investors into a market has the ability to improve the accuracy of public information (Umutlu et al., 2010) and to enhance corporate governance and monitoring (Li et al., 2011). The previous literature also associates the presence of foreign investors with improvements in reporting standards and significant reductions in transaction costs, informational costs and risk exposure (see, e.g., Vo, 2015). Overall, foreign ownership matters, and its presence brings stability to local markets (Wang, 2007; Umutlu et al., 2010; Li et al., 2011; Vo, 2015), which in turn results in lower stock return volatility.

3.3 Instrumental Variables Approach

Societal trust has been shown to promote economic growth and investment and encourage investors' participation in financial markets and financial development (Knack and Keefer, 1997; Zak and Knack, 2001; Guiso et al., 2004, 2008). Previous studies have also suggested that risk is a function of the quality of the institutional environment (i.e., property rights protection, quality of government) (see, e.g., Morck et al., 2000; Jin and Myers, 2006). Although our model specification controls for such factors (see Section 3.4), endogeneity issues could affect our coefficient estimates as unobservable determinants of trust may influence firm risk.

The academic literature contends that institutional quality has the ability to produce societal trust (Paxton, 2002) and societal trust can produce institutional quality (Knack, 2002; Paxton, 2002), indicating a bidirectional version of causality. For example, Aghion et al. (2010)

document a link between trust and government regulation and argue that the public demand for trust is formed by distrust, whereas the formation of trust may be discouraged by regulation. It is important to note that prior research also suggests that individual idiosyncratic features (such as education and religious upbringing (see, e.g., Guiso et al., 2004)) can create considerable differences in the levels of trust across individuals, regions and countries. This view is also supported by a survey from Sapienza and Zingales (2012), which indicates that during the last months of 2008, people became less confident (decrease in trust) in investing in the financial markets due to government intervention.

To tackle the potential endogeneity issue, an instrumental variables approach based on two-stage least squares (2SLS) is utilized. Previous studies suggest that racial heterogeneity is associated with a reduction in the levels of bilateral trust (Gupta et al., 2018). Leigh (2006) supports an inverse relationship between ethnic diversity and trust on the basis that diverse communities find it more challenging to enforce a system of social norms and thus trusting one another becomes less likely. Given that a country's ethnicity and diversity are sticky over long periods (e.g., Guiso et al., 2006), the effect of racial heterogeneity is quite likely to be exogenous.

To capture the racial heterogeneity in the shareholder's country, we use ethnic fractionalization data as described in Alesina et al. (2003), specifically the ethnic fractionalization of the country of origin of the largest shareholder (SHETHFR). We anticipate a negative impact of SHETHFR on trust, as Alesina et al. (2003) suggest that homogeneous communities (such as Japan) have higher levels of social capital since they experience higher levels of social interaction. In addition, Glaeser et al. (2000) and Knack and Keefer (1997) report that racial factors may affect the level of trustworthiness among individuals. Although we anticipate a negative relationship between SHETHFR and SHTRUST, it is unlikely that this will influence firm risk.

3.4 Empirical Model

To test our hypothesis, we use the following model specification:

$$RISK = \beta_0 + \beta_1 SHTRUST + \beta_2 ROA + \beta_3 ln(ASSETS) + \beta_4 LEVERAGE + \beta_5 MB +$$

$$\beta_6 ln(AGE) + \beta_7 BIG4 + \beta_8 GDPGR + \beta_9 INVPROT + \sum YEAR + \sum INDUSTRY + \epsilon$$
(2)

The dependent variable is a vector (RISK) representing the two measures of firm risk, as presented in Section 3.1, and SHTRUST represents our measure of the societal trust of the country of origin of the largest shareholder, as presented in Section 3.2. Several firm-level variables are employed to control cross-sectional differences in firm characteristics that may influence risk. We control for profitability and size using the ratio of earnings before interest and taxes (EBIT) to total assets (ROA) and the natural logarithm of total assets (ln(ASSETS)), respectively, since larger and more profitable firms are more likely to experience lower return volatility (Pastor and Veronesi, 2003). As a measure of financial leverage, we use the ratio of total debt over total assets (LEVERAGE) because financially distressed firms are more likely to be leveraged and have higher return volatility (Rajgopal and Venkatachalam, 2011).

Previous studies provide an association between volatility and firm growth opportunities (Cao et al., 2008) and establish firm age as an important determinant of volatility since younger firms have more volatile cash flows than older firms (Pastor and Veronesi, 2003; Cao et al., 2008). We further account for firm growth opportunities, operationalized as the market-to-book value of equity (MB), and for firm age (ln(AGE)), measured as the natural logarithm of 1 plus the number of years since incorporation. We employ an indicator (BIG4) to control for the firm being audited by a Big-4 audit company (Deloitte, PricewaterhouseCoopers, Ernst & Young, and KPMG) as Big-4 audit firms are associated with lower return volatility (Gul et al., 2010). Considering the established link between market returns and GDP and investor protection (see, e.g., Morck et al., 2000), we conclude our model

for country-level controls and incorporate the annual growth in GDP (GDPGR) and the investor protection index (INVPROT). Following Pevzner et al. (2015), we construct INVPROT as the sum of the anti-self-dealing index from Djankov et al. (2008) and the rule of law index from Kaufmann and Kraay (2017) after standardizing both indices to be between zero and one. The standard errors of all the regression estimates are adjusted using heteroskedasticity-corrected and clustered robust standard errors, clustered on firms. Finally, we control for year and industry dummies and winsorize continuous variables at the top and bottom 1% to mitigate the effect of outliers; we present the variable definitions in Appendix A.

3.5 Data

Our sample spans a 17-year period from 2000 to 2016. We begin our sample construction process with the entire universe of active and listed firms in the DataStream database, which is our primary source for stock price information and accounting data (e.g., similar to Ferreira and Matos, 2008; Pevzner et al., 2015). Next, we collect ownership data using Thomson Reuters Eikon.⁴ One of the main advantages of this source is the availability of data at the investor level, a property which enables us to not only identify each company's blockholders, and subsequently the controlling shareholder, but also to identify shareholders originating from a country different to the country of the company headquarters.⁵ We also

⁴ Thomson Reuters Eikon is used because it firstly, provides global coverage for ownership holdings across more than 70,000 securities in 70+ markets

⁽see http://www.refinitiv.com/content/dam/marketing/en_us/documents/fact-sheets/ownership-profiles-fact-sheet.pdf), and has been used in prior studies (e.g., Anand et al., 2019; Chahine et al., 2020; Chantziaras et al., 2020). Second, ownership data through Thomson Reuters Eikon share similar properties with other databases employed in the literature (e.g., the Orbis, Amadeus and Osiris databases available through Bureau Van Dijk (e.g., Li et al., 2011; Bena and Ortiz-Molina, 2013; Lin et al., 2013; Cumming et al., 2019)), and it differs from the Fact-Set/LionShares database as the latter specialises in global institutional ownership (e.g., Ferreira and Matos, 2008; Bena et al., 2017).

⁵ Through Thomson Reuters Eikon, data can be obtained at the investor level, indicating a) the full name of the investor; b) a description of the type of investor (e.g., institution, individual, entity, etc.); c) the investor's country (i.e. country of origin or headquarters for individuals and entities, respectively); and d) the exact percentage of stocks held.

include the country-level and societal trust data, available through the World Bank and the World Values Survey (WVS), respectively.

[Insert Table 1 about here]

The starting point of our sample period begins in 2000 due to poor coverage of the ownership structure data available through Eikon. We begin with 5,243 listed firms with common support from the Eikon and DataStream databases, which are controlled by a foreign shareholder and for which the country of origin of the largest shareholder is covered by the WVS. Our data requirements for the stock price information and the control variables for our main model (2) further drop 337 and 871 firms, respectively, due to missing data. Following previous studies (e.g., Beck et al., 2013), our sample selection criteria require at least two firm-year observations for each firm within one country and at least four firms in one country, and thus we eliminate 1,240 firms. Our final sample comprises 2,795 firms controlled by foreign shareholders (translated into 12,496 observations; see Table 1 for a description) scattered across 43 countries (see Table 2).

[Insert Table 2 about here]

In the right part of Table 2, we present the average value of our trust measure in each country. We observe that the Philippines, Colombia, Brazil, Peru, and Malaysia are among the bottom five countries in terms of societal trust (measured at the country of corporate headquarters – TRUST) as less than 10% of respondents of the WVS answered that "Most people can be trusted". On the other hand, Norway, Denmark, Sweden, and Finland belong to the top four countries as more than 58% of their citizens indicated that people in their country can be trusted. Another interesting pattern in our data is the level of trust of the largest shareholder (SHTRUST). We observe that Russia, Argentina, Peru, Chile, and Indonesia are among the bottom five countries in terms of shareholder country trust. In contrast, Finland,

New Zealand, Sweden, Norway, the Philippines, and Belgium attract shareholders from countries with higher levels of societal trust as more than 40% of the citizens of the country of origin of the largest shareholder indicate that people in their country can be trusted.

4. Empirical Results

4.1 Univariate Analysis

The descriptive statistics for the main variables used in our analysis are provided in Table 3. The first two lines present a description of the dependent variables, followed by the measure of shareholder country trust and the rest of the control variables. The descriptive statistics indicate that the average value of shareholder country trust is 35.7%. The average firm has a leverage ratio of 0.576, a market-to-book ratio of 1.907, and exhibits negative profitability, as suggested by ROA, which is on average -17.9%. Additionally, more than half of our in-sample firms are audited by a Big-4 audit firm (58.1%). We also present the mean-variance inflation factors (VIFs) in each model and show evidence that multicollinearity is not likely to influence our results.

[Insert Table 3 about here]

4.2 Multivariate Analysis

We conduct a multivariate analysis to investigate any association between the level of shareholder country trust and firm risk. Table 4 presents the results of the OLS and 2SLS analyses, whereby we analyse the impact of foreign shareholder's trust on a firm's total and idiosyncratic risk (Columns 1 and 3 and 2 and 4, respectively). We include OLS estimates to facilitate benchmarking for the 2SLS estimates (Columns 3 and 4). We also provide first-stage results in Appendix B (for the sake of brevity, we do not report the first-stage results). We observe that the coefficient of SHTRUST is negative and statistically significant, after

controlling for numerous firm-level and country-level control variables, supporting our hypothesis that foreign shareholder trust reduces firm risk (either for the OLS – Columns 1 and 2 - or the 2SLS estimates – Columns 3 and 4). In all models, the mean VIFs are around 1.3, implying no multicollinearity. The statistically significant values of the Hausman tests suggest that the hypothesis of no endogeneity cannot be rejected. The partial *R*-squares and the *F*-statistics indicate that the instrument is highly correlated with the endogenous variable, while the *F*-statistics are above the threshold of 10 (Staiger and Stock, 1997), indicating a strong instrument. In terms of economic significance and drawing upon the OLS estimates, a one standard deviation change in SHTRUST (0.121) is associated with a 3.61% (3.78%) decrease in firm idiosyncratic (total) risk.

[Insert Table 4 about here]

Referring to the control variables, the negative coefficients of ROA and ln(ASSETS) support the notion that more profitable and larger firms will experience lower return volatility (Pastor and Veronesi, 2003). Our evidence also suggests that risk reduces with firm age, which is aligned with the findings of Pastor and Veronesi (2003) and Cao et al. (2008). Finally, the negative and highly significant coefficient of BIG4 verifies the documented negative association between Big-4 auditors and lower return volatility (Gul et al., 2010).

5. Formal Institutions

5.1 Cross-Country Variation in the Effect of Trust on Firm Risk

The existing literature indicates that informal institutions, including social norms, religiosity and trust, will influence corporate behaviour differently in different institutional settings (North, 1994; Guiso et al., 2004). In particular, informal institutions will be more important in regions where formal institutions are less effective and will act as substitutes to formal institutions (see e.g., North, 1994; Brockman et al., in press). This notion is empirically

supported by the evidence provided by Guiso et al. (2004) for Italy and by Ang et al. (2015) for China. Recent evidence from the banking sector indicates that informal institutions play an important role in loan volume and loan conditions (Qian et al., 2018), with their effect being more prominent in weak institutional contexts and in contexts with weak legal protection or law enforcement. Against this background, we anticipate that shareholders' trust will vary across countries due to differences in their institutional environment. To this end, we empirically test the effect of foreign shareholder trust on firm risk for firms located in 1) countries with low investor protection (Subsection 5.1.1) and 2) weak institutional contexts (Subsection 5.1.2).

5.1.1 Poor Investor Protection, Foreign Shareholder Trust and Firm Risk

As we highlighted previously, informal institutions matter more in the absence of formal institutions (North, 1994; Guiso et al., 2004), and their effect is expected to be more pronounced in regions where formal institutions are less effective, and so will act as substitutes to formal institutions (North, 1994). Pevzner et al. (2015) provide evidence that the effect of trust on investors' reactions to corporate financial disclosures is more pronounced in countries with low investor protection, a finding which is consistent with the law and finance literature. To empirically test for cross-country variations in the effect of foreign shareholder trust on firm risk, we separate our sample using the sample median of INVPROT in each year.⁶ We present the results in Table 5, where we observe that SHTRUST is negative and statistically significant in countries with low investor protection (see Columns 1 and 2 of Table 5), while it lacks statistical significance in countries with higher investor protection (see Columns 3 and 4 of Table 5). We also test for homogeneity in the pairwise estimated coefficients of SHTRUST,

⁶ Following Brockman et al. (in press), we consider two alternative measures of investor protection, namely the rule of law (RULAW) and the control for corruption (CORRUP) indices available through the World Bank. Repeating our analyses after separating our sample using the sample median of either of the two aforementioned indices yields similar results.

using Wald tests, and find that their difference is statistically significant at the 1% level (Columns 1 and 3 *t*-stat = 20.723 and Columns 2 and 4 *t*-stat = 12.33) across countries with both low and high investor protection. This evidence indicates that the impact of foreign shareholder trust on market risk is more pronounced in countries with poor investor protection.

[Insert Table 5 about here]

5.1.2 Weak Institutional Contexts, Foreign Shareholder Trust and Firm Risk

Aside from cross-country variations in investor protection, we also investigate whether the effect of foreign shareholder trust varies across different institutional contexts. Less developed contexts constitute a challenging institutional environment due to imperfect contracting, less- or under-developed market mechanisms, unpredictable and burdensome regulations, bureaucratic procedures, political instability or discontinuity in government policies (Khanna and Palepu, 1997, 2000). These authors highlight that such contexts face various market failures, attributable to a lack of adequate disclosure and weak corporate governance and control. In addition, another strand of the literature underlines the inefficiencies in the judiciary system, especially in terms of law enforcement (see, e.g., Qian et al., 2018), and shows that even after laws and regulations have been established, their enforcement often remains poor (Pistor et al., 2000). Therefore, the role of informal institutions becomes more important in weak institutional contexts. Under certain circumstances (e.g., with the existence of imperfections in informal institutions) these informal frameworks can even replace formal institutions (Allen et al., 2005).

To empirically investigate whether there are any variations in the effect of foreign shareholder trust on firm risk, we separate our sample into weak/strong institutional contexts (i.e., developing vs. developed countries) using the United Nation's classification system (https://unstats.un.org/unsd/methodology/m49/#fn6).⁷

[Insert Table 6 about here]

Table 6 presents the results of our analyses. We observe that the effect of foreign shareholder trust is more prominent in weak institutional contexts as SHTRUST is negative and statistically significant at the 1% level of significance (see Columns 3 and 4 of Table 6), while it lacks statistical significance in strong institutional contexts (see Columns 1 and 2 of Table 6). Additionally, the Wald tests for homogeneity in pairwise coefficients indicate that the differences in the coefficients of SHTRUST are statistically significant at 1% (Columns 1 and 3 *t*-stat = 10.7 and Columns 2 and 4 *t*-stat = 9.005) between weak and strong institutional contexts. This evidence indicates that the impact of foreign shareholder trust on market risk varies with the strength of the institutional context. The results presented in this and the previous subsections corroborate previous studies and show that informal institutions have larger effects in regions where formal institutions are less effective (North, 1994; Guiso et al., 2004; Qian et al., 2018). More broadly, these results add to existing evidence that demonstrates informal institutions (such as trust) have the ability to constrain firm risk (see, e.g., Kanagaretnam et al., 2017; Hasan and Habib, 2019).

5.2 Investors' Horizon and the Effect of Foreign Shareholder Trust on Firm Risk

In addition to the cross-country variations, we also investigate whether foreign shareholder behaviour varies according to their investment horizon. The existing literature suggests that short-term investors are less likely to actively monitor firms since they possess

⁷ Following Brockman et al. (in press), we consider two alternative measures to capture the quality of the institutional environment, namely the voice and accountability (VOICACC) and the government effectiveness (GOVEFF) indices available through the World Bank. Repeating our analyses after separating our sample using the sample median of either of the two aforementioned indices, yields similar results.

informational advantages that they use to trade regularly (e.g., Bushee and Goodman, 2007; Brockman and Yan, 2009). On the other hand, long-term investors more actively monitor firms as they have long-term interests in firms and often invest large stakes (Bushee, 1998; Chen et al., 2007). Against this background, we draw upon relevant studies (e.g., Harford et al., 2018) and classify foreign major shareholders as per their investment horizon (short-term vs. longterm). We incorporated the suggested papers on page 27:

Döring et al. (2020) summarize the debate concerning the relevance of shareholder investment horizon in influencing corporate governance decisions, while Ding et al. (2020) investigate the information asymmetries that exist between local and foreign institutional investors. In the latter study, local institutional investors were found to have an information advantage over foreign institutional investors when investing in partially state-owned Chinese enterprises, although this local information advantage was not present when there was no state ownership. More specifically, we start at the investor level and measure the horizons of investors based on their portfolio turnover. For every investor and every year, we look back three years (similar to Harford et al., 2018) and compute the fraction of the investor's portfolio that is no longer held at the end of the period. We then classify investors as having either a short-term or long-term horizon, using a portfolio turnover cut-off of 35% (similar to Froot et al., 1992; Harford et al., 2018).

[Insert Table 7 about here]

In Table 7 we separate investors according to their investment horizon, namely long term (Columns 1 and 2) and short term (Columns 3 and 4). We observe that SHTRUST attains negative coefficients across all models, but it is statistically significant (at 1%) only across foreign major shareholders with a long-term investment horizon (see Columns 1 and 2 of Table 7). Further evidence from the Wald test confirms that the difference in the pairwise estimated coefficients is statistically significant at the 1% level of significance (Columns 1 and 3 *t*-stat =

22.837 and Columns 2 and 4 *t*-stat = 18.43). The results suggest that foreign shareholder trust has a more prominent effect in reducing firm market risk across firms controlled by investors with a long-term horizon.

5.3 Difference between Foreign Shareholder Trust and Local Trust

In this section, we investigate whether international investors originating from more trusting environments, compared to the country of corporate headquarters, are capable of bringing stability to the market and thus, further reduce firm risk (Umutlu et al., 2010; Li et al., 2011). To accommodate this effect, we repeat our analyses and the extensions (as presented in sub-sections 5.1.1, 5.1.2, and 5.2) using the difference between the level of trust in the country of origin of the largest shareholder and the country of corporate headquarters (TRUSTDIFF, where positive values indicate that the shareholder comes from a more trusting environment) as the main independent variable. This is consistent with previous studies (see, e.g., Ahern et al., 2015).

[Insert Table 8 about here]

Table 8 presents the results of our analyses, which are qualitatively similar to our previous analyses. More specifically, we observe that the TRUSTDIFF coefficients are negative and statistically significant at 1%, suggesting that the higher the difference between shareholder trust and local trust, the higher the reduction of firm risk (see Columns 1 and 2 of Table 8). This effect is amplified in regions where formal institutions are less effective (i.e., poor investor protection and weak institutional contexts; see Columns 3 and 4 and Columns 9 and 10 of Table 8, respectively) or among firms controlled by foreign investors with a long-term horizon (see Columns 11 and 12 of Table 8). Collectively, these results indicate that the effect of international investors originating from more trusting environments is an important determinant of firm risk and its effect varies with the effectiveness of the institutional context.

6. Sensitivity Analyses

Several robustness checks are also undertaken to assess the validity of the earlier reported results. First, we assess whether our results are sensitive to changes in the controlling shareholder. As discussed in our theoretical development, the negative impact of foreign shareholder trust on risk stems from large shareholders' decision to hold corporate shares for longer (inaction). As a consequence, one could claim that firms controlled by shareholders from high-trust environments are less likely to exhibit changes in control as these investors are more likely to "stick" to the firm, and thus the observed effect is due to a long-term relationship between the shareholder and the firm. Indicatively, 27% of our in-sample firms have experienced changes in their controlling shareholder.

To capture this effect, we conduct further analyses: a) we augment our model specification with an indicator variable that equals one if the current shareholder is not the same as in the previous period (SHCHANGE, see Columns 1 and 2 of Table 9); and b) we partition our sample into firms where the new controlling shareholder comes from a more (less) trusting environment as compared to the previous shareholder (Columns 3 and 4 (5 and 6) of Table 9).

The results indicate that changes in the controlling shareholder increase firm risk as the SHCHANGE attains a positive and statistically significant coefficient at 1%. Partitioning the sample reveals that the effect of foreign shareholder trust is significant only when the new controlling shareholder comes from a more trusting environment, while has no effect if they originate from a country with lower trust as compared to the previous controlling shareholder (in year *t*-1).

[Insert Table 9 about here]

Second, we perform a Heckman-type correction to address potential selection bias (similar to El Ghoul et al., 2017). First, we start with the entire DataStream universe and

estimate a probit model (first stage) to determine if the firm belongs in our sample against firmlevel covariates (i.e., size, age, leverage, growth opportunities, alongside with year, country, and industry effects to control for unobservable time, country, and industry factors, respectively). Next, we predict the inverse Mills ratio (LAMBDA) and include it as an additional control in our main analysis to control for selection bias. Importantly, this analysis demonstrates that the original results remain unchanged. The additional documentation and results from this and all subsequent tests are available in the online appendix.

Third, and to further support the primary findings, alternative measures of firm risk are applied. This approach applies a simple model market regression (without the Fama and French (1993) three factors) and the subsequent recalculation of idiosyncratic risk (IRISKMM). The logarithmic transformation of our risk measures (ln(TRISK) and ln(IRISK)), similar to Pevzner et al. (2015), is also applied to address the concern that the original measure has a skewed distribution. In line with prior studies (e.g., Sila et al., 2016), we also consider proxies of firm risk based on operating performance and calculate the three- (five-) year standard deviation of ROA and ROE, namely the SD3ROA and SD3ROE (SD5ROA and SD5ROE). Fourth, we follow Kanagaretnam et al. (2019) and use the TRUST_INDEX (Medrano, 2011) as an alternative measurement of foreign shareholder trust. Fifth, we assess whether our results are sensitive to the selection of our instrument. Following Brockman et al. (in press), we repeat our 2SLS analyses using the genetic distance between the population of the country of the firm headquarters and that of the origin of the foreign shareholder (GENETDIST).⁸ Sixth, we rerun our analysis after excluding the financial crisis period of 2007-2009. This exclusion is because the prior literature associates the crisis period with an increase in idiosyncratic return volatility (Chichernea et al., 2015). Seventh, we examine the sensitivity of our results across different

⁸ GENDIST captures the probability that random selection by two alleles at a given locus from the population of one country and the population of the other country will be different (Spolaore and Wacziarg (2009). Data obtained from Enrico Spolaore's website (http://sites.tufts.edu/enricospolaore/category/personal-webpage/).

thresholds of foreign controlling shareholders, namely at 10%, 20% and 25%. Eighth, we conduct additional sensitivity tests in which we a) control for country FEs and use heteroskedasticity-corrected robust standard errors (clustered on firms); b) use heteroskedasticity-corrected robust standard errors clustered on countries; and c) control for country FEs and use heteroskedasticity-corrected robust standard errors clustered on countries. The results are qualitatively similar to the main analysis, verifying that foreign shareholder trust has a negative impact on firm market risk, even with the alternative definitions of the dependent variables; with the use of alternative measures of foreign shareholder trust; alternative instruments employed; different thresholds of foreign block-ownership; with the inclusion of country fixed effects; and using different clusters in standard errors.

A number of other variables that have been found, or suggested (explicitly or implicitly), to influence a stock's systematic risk are now investigated. These include operating leverage (OPLEV), measured as the ratio of net property, plant and equipment to total assets (Lev, 1974; Mandelker and Rhee, 1984); dividend pay-out ratio (DIVPAYOUT), measured as the ratio of common dividends divided by income before extraordinary items as a proxy for risk (Pastor and Veronesi, 2003; Hoberg and Prabhala, 2009); research and development expenditure (RDEXP) and capital expenditure (CAPEX), both divided by total assets, as proxies for investment and growth opportunities (Guay, 1999); the standard deviation of cash flows from operation (STDCF) scaled by total assets for the last three years (Hasan and Habib, 2019); and an indicator variable that equals 1 if the firm is listed in more than one stock exchange (CROSSLISTED). Additionally, market competition is controlled using the Herfindahl index (HHI) since Irvine and Pontiff (2009) show that firm competition affects idiosyncratic risk by increasing cash flow variability. We also augment the model for ownership structure variables, such as the stakes owned by foreign (FOR_OWN) and

institutional (INST_OWN) investors. The incorporation of all the above variables does not change the core results.

We further assess the effect of foreign shareholder trust on firm risk after controlling for cultural effects, and we augment the model with Hofstede's (2001) cultural dimensions to capture individualism, power distance, uncertainty avoidance, and masculinity measured either at the country of firm headquarters or the country of origin of the foreign shareholder. We also include country-pair characteristics that might affect the level of trust between investors from two different countries, namely the distance between each firm's country of headquarters and the country of origin of the foreign shareholder (GEODIST), as well as an indicator variable that equals 1 if the country of the firm's headquarters and the country of origin of the foreign shareholder share borders (SHAREBOARDER).⁹ We also augment the model for country-level variables by including stock market development (measured as the total market capitalization divided by gross domestic product, MCAPCNT) and a country's legal origin (indicator variables for common law countries, COMLAW). Additionally, we control for the GINI coefficient of each country of corporate headquarters to capture any income equalities and for each country's credit rating of sovereign debt (i.e. using credit ratings through Standard and Poor's, Fitch, and Moody's). Adding these variables to the model does not affect our results. Finally, we probe the robustness of our main results after excluding firms from the US, Japan, India, Canada and Australia. Doing so addresses concerns that the results are driven by the large representation of these countries. This final analysis still confirms that foreign shareholder trust continues to have a significant negative effect on our two measures of risk.

⁹ We obtain the latitude and longitude data between the two points from Google Maps (https://developers.google.com/public-data/docs/canonical/countries_csv) and employ the Harvershine formula to calculate the distance. $DIS = R \times 2 \times \arcsin(\min(1, \sqrt{a}))$, where R =approximately 6,378 kilometres (radius of the earth) and $a = (sin(dlat/2))^2 + cos(lat1) \times cos(lat2) \times (sin(dlon/2))^2$. In the latter expression dlat = lat2 - lat1 and dlon = lon2 - lon1, where lat1 and lon1 are the latitude and longitude of the firm location and lat2 and lon2 are the latitude and longitude of the location of the foreign shareholder.

This evidence provides further support for the hypothesis that a higher level of foreign shareholder trust reduces firm risk.

7. Conclusion

In this study, we highlight the role that foreign shareholders' trust has in mitigating firm risk. The empirical evidence from this study is consistent with a significant negative relationship between foreign shareholder trust and market risk. This relationship remains robust after undertaking a range of sensitivity analyses and after considering potential endogeneity and selection bias issues. The effect of trust also deviates with cross-country and investor horizon variations. In particular, the effect becomes more pronounced for firms located in countries with low investor protection and a weaker institutional environment and for firms controlled by investors with a long-term horizon. We further demonstrate that the effect of international investors originating from more trusting environments is an important determinant of firm risk, and the effect varies with the effectiveness of the institutional context.

Our results contribute to an emerging stream of the corporate finance literature that investigates the effects of trust on corporate matters (see, e.g., Pevzner et al., 2015; Bottazzi et al., 2016; Cingano and Pinotti, 2016; Dudley and Zhang, 2016; Qian et al., 2018; Hasan and Habib, 2019). In addition, a significant negative relationship between foreign shareholder trust and risk, highlights the role that trust plays as a factor that may mitigate excessive corporate risk-taking.

This study is one of the first to consider the impact of the trust of large foreign shareholders on corporate decision making. In summary, the paper makes the following contributions to this literature: First, this investigation documents additional empirical insights into how trust affects corporate market risk; Second, a multi-country perspective, allows consideration of more complex legal and property rights factors; Third, measuring total and

34

idiosyncratic risk in a cross-country setting provides a better understanding of the overall effect of trust on market risk; Fourth, consideration of different foreign shareholder investment horizons allows investigation of the role that long-term oriented foreign shareholders from high-trust countries play in reducing firm market risk.

In conclusion, the findings show that after considering firm and country-level characteristics, and controlling for industry and year effects, foreign controlling shareholders' trust reduces firm market risk. Firms controlled by large foreign shareholders, originating from high-trust environments, display lower levels of marker risk. Trust is also more important for firms located in countries with a less favourable, or effective, institutional environment, suggesting that trust may support the institutional environment. Finally, the geographic location where the controlling shareholders are based and its degree of societal trust matters in determining firm risk.

This paper also has important policy implications for international business in that corporations could be encouraged to attract foreign shareholders from more trusting countries to mitigate their market risk, especially for those domiciled in countries where there is a weak institutional environment. To support this position, policy makers could design policies that encourage portfolio or direct investment from high trusting countries to mitigate the market risk of domestic firms. One limitation of this study is that risk is measured at the country level. Consequently, future studies could identify and develop firm-level trust measures to examine the role that trust may play in firm-level corporate decision making. Furthermore, the effects of the economic crisis arising from the COVID-19 epidemic may allow future studies to investigate the extent that trust affects investor behaviour when there are sudden changes to market risk.

35

References

Abdoh, H., Varela, O., 2017. Product market competition, idiosyncratic and systematic volatility. Journal of Corporate Finance. 43, 500-513. http://doi.org/10.1016/j.jcorpfin.2017.02.009

Aghion, P., Algan, Y., Cahuc, P., Shleifer, A., 2010. Regulation and distrust. The Quarterly Journal of Economics. 125, 1015-1049. http://doi.org/10.1162/qjec.2010.125.3.1015

Ahern, K.R., Daminelli, D., Fracassi, C., 2015. Lost in translation? The effect of cultural values on mergers around the world. Journal of Financial Economics. 117, 165-189. http://doi.org/10.1016/j.jfineco.2012.08.006

Alesina, A., Devleeschauwer, A., Easterly, W., Kurlat, S., Wacziarg, R., 2003. Fractionalization. Journal of Economic Growth. 8, 155-194. http://doi.org/10.3386/w9411

Allen, F., 2005. Corporate governance in emerging economies. Oxford Review of Economic Policy. 21, 164-177. http://doi.org/10.1093/oxrep/gri010

Allen, F., Qian, J., Qian, M., 2005. Law, finance, and economic growth in China. Journal of Financial Economics. 77, 57-116. http://doi.org/10.1016/j.jfineco.2004.06.010

Amihud, Y., 2002. Illiquidity and stock returns: Cross-section and time-series effects. Journal of Financial Markets. 5, 31-56. http://doi.org/10.1016/S1386-4181(01)00024-6

Anand, A., Irvine, P., Liu, T., 2019. Does institutional trading affect underwriting? Journal of Corporate Finance. 58, 804-823. http://doi.org/10.1016/j.jcorpfin.2019.101495

Andres, C., 2008. Large shareholders and firm performance - An empirical examination of founding-family ownership. Journal of Corporate Finance. 14, 431-445. http://doi.org/10.1016/j.jcorpfin.2008.05.003

Ang, A., Hodrick, R.J., Xing, Y., Zhang, X., 2006. The cross-section of volatility and expected returns. The Journal of Finance. 61, 259-299. http://doi.org/10.1111/j.1540-6261.2006.00836.x

Ang, J.S., Cheng, Y., Wu, C., 2015. Trust, investment, and business contracting. Journal ofFinancialandQuantitativeAnalysis.50,569-595.http://doi.org/10.1017/S002210901500006X

Arrow, K.J., 1972. Gifts and exchanges. Philosophy & Public Affairs. 1, 343-362. http://doi.org/10.1515/9781400853564.139

Attig, N., Cleary, S., El Ghoul, S., Guedhami, O., 2012. Institutional investment horizon and investment–cash flow sensitivity. Journal of Banking & Finance. 36, 1164-1180. http://doi.org/10.1016/j.jbankfin.2011.11.015

—, 2013. Institutional investment horizons and the cost of equity capital. Financial Management. 42, 441-477. http://doi.org/10.1111/j.1755-053X.2012.01221.x

Baird, I.S., Thomas, H., 1985. Toward a contingency model of strategic risk taking. Academy of Management Review. 10, 230-243. http://doi.org/10.5465/AMR.1985.4278108
Balloch, A., Nicolae, A., Philip, D., 2015. Stock market literacy, trust, and participation. Review of Finance. 19, 1925-1963. http://doi.org/10.1093/rof/rfu040

Beck, T., Demirgüç-Kunt, A., Merrouche, O., 2013. Islamic vs. conventional banking: Business model, efficiency and stability. Journal of Banking & Finance. 37, 433-447. http://doi.org/10.1016/j.jbankfin.2012.09.016

Bena, J., Ferreira, M.A., Matos, P., Pires, P., 2017. Are foreign investors locusts? The long-term effects of foreign institutional ownership. Journal of Financial Economics. 126, 122-146. http://doi.org/10.1016/j.jfineco.2017.07.005

Bena, J., Ortiz-Molina, H., 2013. Pyramidal ownership and the creation of new firms. Journal of Financial Economics. 108, 798-821. http://doi.org/10.1016/j.jfineco.2013.01.009

Bodnaruk, A., Kandel, E., Massa, M., Simonov, A., 2008. Shareholder diversification and the decision to go public. The Review of Financial Studies. 21, 2779-2824. http://doi.org/10.1093/rfs/hhm036

Bottazzi, L., Da Rin, M., Hellmann, T., 2016. The importance of trust for investment: Evidence from venture capital. The Review of Financial Studies. 29, 2283-2318. http://doi.org/10.1093/rfs/hhw023

Brockman, P., El Ghoul, S., Guedhami, O., Zheng, Y., in press. Does social trust affect international contracting? Evidence from foreign bond covenants. Journal of International Business Studies. http://doi.org/10.1057/s41267-020-00348-4

Brockman, P., Yan, X., 2009. Block ownership and firm-specific information. Journal of Banking & Finance. 33, 308-316. http://doi.org/10.1016/j.jbankfin.2008.08.011

Buonanno, P., Montolio, D., Vanin, P., 2009. Does social capital reduce crime? The Journal of Law and Economics. 52, 145-170. http://doi.org/10.1086/595698

Bushee, B.J., 1998. The influence of institutional investors on myopic R&D investment behavior. The Accounting Review. 73, 305-333. http://doi.org/10.2307/248542

——, 2001. Do institutional investors prefer near-term earnings over long-run value? Contemporary Accounting Research. 18, 207-246. http://doi.org/10.1506/J4GU-BHWH-8HME-LE0X

—, 2004. Identifying and attracting the "right" investors: Evidence on the behavior of institutional investors. Journal of Applied Corporate Finance. 16, 28-35. http://doi.org/10.1111/j.1745-6622.2004.00005.x

Bushee, B.J., Goodman, T.H., 2007. Which institutional investors trade based on private information about earnings and returns? Journal of Accounting Research. 45, 289-321. http://doi.org/10.1111/j.1475-679X.2007.00234.x

Bushee, B.J., Noe, C.F., 2000. Corporate disclosure practices, institutional investors, and stock return volatility. Journal of Accounting Research. 38, 171-202. http://doi.org/10.2307/2672914

Cao, C., Simin, T., Zhao, J., 2008. Can growth options explain the trend in idiosyncratic risk? The Review of Financial Studies. 21, 2599-2633. http://doi.org/10.1093/rfs/hhl039

Chahine, S., Colak, G., Hasan, I., Mazboudi, M., 2020. Investor relations and IPO performance. Review of Accounting Studies. 25, 474-512. http://doi.org/10.1007/s11142-019-09526-8

Chantziaras, A., Dedoulis, E., Grougiou, V., Leventis, S., 2020. The impact of religiosity and corruption on CSR reporting: The case of U.S. banks. Journal of Business Research. 109, 362-374. http://doi.org/10.1016/j.jbusres.2019.12.025

Chen, X., Harford, J., Li, K., 2007. Monitoring: Which institutions matter? Journal of Financial Economics. 86, 279-305. http://dx.doi.org/10.1016/j.jfineco.2006.09.005

Chichernea, D.C., Petkevich, A., Zykaj, B.B., 2015. Idiosyncratic volatility, institutional ownership, and investment horizon. European Financial Management. 21, 613-645. http://doi.org/10.1111/j.1468-036X.2013.12033.x

Chircop, J., Johan, S., Tarsalewska, M., 2020. Does religiosity influence venture capital investment decisions? Journal of Corporate Finance. 62, 101589. http://doi.org/10.1016/j.jcorpfin.2020.101589

Cingano, F., Pinotti, P., 2016. Trust, firm organization, and the pattern of comparative advantage. Journal of International Economics. 100, 1-13. http://doi.org/10.1016/j.jinteco.2016.01.006

Claessens, S., Djankov, S., Fan, J.P.H., Lang, L.H.P., 2002. Disentangling the incentive and entrenchment effects of large shareholdings. The Journal of Finance. 57, 2741-2771. http://doi.org/10.1111/1540-6261.00511

Cronqvist, H., Fahlenbrach, R., 2009. Large shareholders and corporate policies. The Review of Financial Studies. 22, 3941-3976. http://doi.org/10.1093/rfs/hhn093

Cumming, D., Peter, R., Sannajust, A., Tarsalewska, M., 2019. Pre-going private ownership around the world. British Journal of Management. 30, 692-711. http://doi.org/10.1111/1467-8551.12281

Cumming, D., Peter, R., Tarsalewska, M., 2020. Public-to-private buyouts and innovation. British Journal of Management. 31, 811-829. http://doi.org/10.1111/1467-8551.12404

Das, T.K., Teng, B.-S., 2004. The risk-based view of trust: A conceptual framework. JournalofBusinessandPsychology.19,85-116.http://doi.org/10.1023/B:JOBU.0000040274.23551.1b

DeMarzo, P., Urošević, B., 2006. Ownership dynamics and asset pricing with a large shareholder. Journal of Political Economy. 114, 774-815. http://doi.org/10.1086/506334

Ding, X., Guedhami, O., Ni, Y., Pittman, J.A., 2020. Local and foreign institutional investors, information asymmetries, and state ownership. Pacific-Basin Finance Journal. 63, 101405. http://doi.org/10.1016/j.pacfin.2020.101405

Djankov, S., La Porta, R., Lopez-de-Silanes, F., Shleifer, A., 2008. The law and economics of self-dealing. Journal of Financial Economics. 88, 430-465. http://doi.org/10.1016/j.jfineco.2007.02.007 Döring, S., Drobetz, W., El Ghoul, S., Guedhami, O., Schröder, H., in press. Institutional investment horizons and firm valuation around the world. Journal of International Business Studies. http://doi.org/10.1057/s41267-020-00351-9

Dorn, D., Huberman, G., 2005. Talk and action: What individual investors say and what they do. Review of Finance. 9, 437-481. http://doi.org/10.1007/s10679-005-4997-z

Dudley, E., Zhang, N., 2016. Trust and corporate cash holdings. Journal of Corporate Finance. 41, 363-387. http://doi.org/10.1016/j.jcorpfin.2016.10.010

Dyer, J.H., Chu, W., 2003. The role of trustworthiness in reducing transaction costs and improving performance: empirical evidence from the United States, Japan, and Korea. Organization Science. 14, 57-68. http://doi.org/10.1287/orsc.14.1.57.12806

Edmans, A., 2014. Blockholders and corporate governance. Annual Review of Financial Economics. 6, 23-50. http://doi.org/10.1146/annurev-financial-110613-034455

El Ghoul, S., Guedhami, O., Kim, Y., 2017. Country-level institutions, firm value, and the role of corporate social responsibility initiatives. Journal of International Business Studies. 48, 360-385. http://doi.org/10.1057/jibs.2016.4

El Ghoul, S., Guedhami, O., Kwok, C.C.Y., Zheng, Y., 2019. Collectivism and the costs of high leverage. Journal of Banking & Finance. 106, 227-245. http://doi.org/10.1016/j.jbankfin.2019.06.004

Faccio, M., Lang, L.H.P., 2002. The ultimate ownership of Western European corporations. Journal of Financial Economics. 65, 365-395. http://dx.doi.org/10.1016/S0304-405X(02)00146-0

Faccio, M., Marchica, M.-T., Mura, R., 2011. Large shareholder diversification and corporate risk-taking. The Review of Financial Studies. 24, 3601-3641. http://doi.org/10.1093/rfs/hhr065

Fama, E.F., French, K.R., 1993. Common risk factors in the returns on stocks and bonds. Journal of Financial Economics. 33, 3-56. http://dx.doi.org/10.1016/0304-405X(93)90023-5

Ferreira, M.A., Matos, P., 2008. The colors of investors' money: The role of institutional investors around the world. Journal of Financial Economics. 88, 499-533. http://doi.org/10.1016/j.jfineco.2007.07.003

Fiegenbaum, A., Thomas, H., 1988. Attitudes toward risk and the risk-return paradox: Prospect theory explanations. The Academy of Management Journal. 31, 85-106. http://doi.org/10.2307/256499

Froot, K.A., Perold, A.F., Stein, J.C., 1992. Shareholder trading practices and corporate investment horizons. Journal of Applied Corporate Finance. 5, 42-58. http://doi.org/10.1111/j.1745-6622.1992.tb00489.x

Gambetta, D. 1988. Can we trust trust? In: Gambetta, D. (Ed), Trust: Making and breaking cooperative relations. Basil Blackwell: New York, NY

García-Kuhnert, Y., Marchica, M.-T., Mura, R., 2015. Shareholder diversification and bank risk-taking. Journal of Financial Intermediation. 24, 602-635. http://doi.org/10.1016/j.jfi.2015.03.001

Georgarakos, D., Pasini, G., 2011. Trust, sociability, and stock market participation. Review of Finance. 15, 693-725. http://doi.org/10.1093/rof/rfr028

Glaeser, E.L., Laibson, D.I., Scheinkman, J.A., Soutter, C.L., 2000. Measuring trust. The Quarterly Journal of Economics. 115, 811-846. http://doi.org/10.1162/003355300554926

Goergen, M., Renneboog, L., 2001. Investment policy, internal financing and ownership concentration in the UK. Journal of Corporate Finance. 7, 257-284. http://doi.org/10.1016/S0929-1199(01)00022-0

Goetzmann, W.N., Kumar, A., 2008. Equity portfolio diversification. Review of Finance. 12, 433-463. http://doi.org/10.1093/rof/rfn005

Goyal, A., Santa-Clara, P., 2003. Idiosyncratic risk matters! The Journal of Finance. 58, 975-1007. http://doi.org/10.1111/1540-6261.00555

Guay, W.R., 1999. The sensitivity of CEO wealth to equity risk: An analysis of the magnitude and determinants. Journal of Financial Economics. 53, 43-71. http://doi.org/10.1016/S0304-405X(99)00016-1

Guiso, L., Sapienza, P., Zingales, L., 2004. The role of social capital in financial development. The American Economic Review. 94, 526-556. http://doi.org/10.1257/0002828041464498

—, 2006. Does culture affect economic outcomes? Journal of Economic Perspectives. 20, 23-48. http://doi.org/10.1257/jep.20.2.23

—, 2008. Trusting the stock market. The Journal of Finance. 63, 2557-2600. http://doi.org/10.1111/j.1540-6261.2008.01408.x

Gul, F.A., Kim, J.-B., Qiu, A.A., 2010. Ownership concentration, foreign shareholding, audit quality, and stock price synchronicity: Evidence from China. Journal of Financial Economics. 95, 425-442. http://dx.doi.org/10.1016/j.jfineco.2009.11.005

Gupta, A., Raman, K., Shang, C., 2018. Social capital and the cost of equity. Journal of Banking & Finance. 87, 102-117. http://doi.org/10.1016/j.jbankfin.2017.10.002

Gürsoy, G., Kürşat, A., 2002. Equity ownership structure, risk taking, and performance: an empirical investigation in Turkish listed companies. Emerging Markets Finance and Trade. 38, 6-25

Harford, J., Kecskés, A., Mansi, S., 2018. Do long-term investors improve corporate decision making? Journal of Corporate Finance. 50, 424-452. http://doi.org/10.1016/j.jcorpfin.2017.09.022

Hasan, M.M., Habib, A., 2019. Social capital and idiosyncratic return volatility. Australian Journal of Management. 44, 3-31. http://doi.org/10.1177/0312896217717573

Hoberg, G., Prabhala, N.R., 2009. Disappearing dividends, catering, and risk. The Review of Financial Studies. 22, 79-116. http://doi.org/10.1093/rfs/hhn073

Hofstede, G.H., 2001. Culture's consequences: Comparing values, behaviors, institutions and organizations across nations. Sage Publications: Thousand Oaks, CA

Inglehart, R. 1999. Trust, well-being and democracy. In: Warren, M.E. (Ed), Democracy and Trust. Cambridge University Press: New York, NY

Irvine, P.J., Pontiff, J., 2009. Idiosyncratic return volatility, cash flows, and product market competition. The Review of Financial Studies. 22, 1149-1177. http://doi.org/10.1093/rfs/hhn039

Jensen, M.C., Meckling, W.H., 1976. Theory of the firm: Managerial behavior, agency costs and ownership structure. Journal of Financial Economics. 3, 305-360. http://doi.org/10.1016/0304-405x(76)90026-x

Jin, L., Myers, S.C., 2006. R2 around the world: New theory and new tests. Journal of Financial Economics. 79, 257-292. http://doi.org/10.1016/j.jfineco.2004.11.003

Jordà, Ò., Schularick, M., Taylor, A.M., 2013. When credit bites back. Journal of Money, Credit and Banking. 45, 3-28. http://doi.org/10.1111/jmcb.12069

Kanagaretnam, K., Lee, J., Chee Yeow, L., Lobo, G.J., 2017. Effects of informal institutions on the relationship between accounting measures of risk and bank distress. Journal of International Accounting Research. 16, 37-66. http://doi.org/10.2308/jiar-51827

Kanagaretnam, K., Lee, J., Lim, C.Y., Lobo, G., 2018. Societal trust and corporate tax avoidance. Review of Accounting Studies. 23, 1588-1628. http://doi.org/10.1007/s11142-018-9466-y

Kanagaretnam, K., Lobo, G.J., Wang, C., Whalen, D.J., 2019. Cross-country evidence on the relationship between societal trust and risk-taking by banks. Journal of Financial and Quantitative Analysis. 54, 275-301. http://doi.org/10.1017/S0022109018000455

[dataset] Kaufmann, D., Kraay, A., 2017. The worldwide governance indicators (WGI) project. The World Bank. http://info.worldbank.org/governance/wgi/#home

Khanna, T., Palepu, K., 1997. Why focused strategies may be wrong for emerging markets. Harvard Business Review. 75, 41-51

—, 2000. The future of business groups in emerging markets: Long-run evidence from Chile. The Academy of Management Journal. 43, 268-285. http://doi.org/10.2307/1556395

Knack, S., 2002. Social capital and the quality of government: Evidence from the States. American Journal of Political Science. 46, 772-785. http://doi.org/10.2307/3088433

Knack, S., Keefer, P., 1997. Does social capital have an economic payoff? A cross-country investigation. The Quarterly Journal of Economics. 112, 1251-1288. http://doi.org/10.1162/003355300555475 La Porta, R., Lopez-de-Silanes, F., Shleifer, A., 1999. Corporate ownership around the world. The Journal of Finance. 54, 471-517. http://doi.org/10.1111/0022-1082.00115

La Porta, R., Lopez-De-Silanes, F., Shleifer, A., Vishny, R., 1997. Trust in large organizations. The American Economic Review. 87, 333-338. http://doi.org/10.3386/w5864

Laeven, L., Levine, R., 2008. Complex ownership structures and corporate valuations. The Review of Financial Studies. 21, 579-604. http://doi.org/10.1093/rfs/hhm068

Leigh, A., 2006. Trust, inequality and ethnic heterogeneity. Economic Record. 82, 268-280. http://doi.org/10.1111/j.1475-4932.2006.00339.x

Lesmeister, S., Limbach, P., Goergen, M., 2018. Trust and shareholder voting. European Corporate Governance Institute, ECGI Working Paper No 569/2018. http://ecgi.global/sites/default/files/working_papers/documents/finallesmeisterlimbachgoerge n.pdf

Lev, B., 1974. On the association between operating leverage and risk. Journal of Financial and Quantitative Analysis. 9, 627-641. http://doi.org/10.2307/2329764

Levine, R., Lin, C., Xie, W., 2018. Corporate resilience to banking crises: The roles of trust and trade credit. Journal of Financial and Quantitative Analysis. 53, 1441-1477. http://doi.org/10.1017/S0022109018000224

Li, D., Moshirian, F., Pham, P.K., Zein, J., 2006. When financial institutions are large shareholders: The role of macro corporate governance environments. The Journal of Finance. 61, 2975-3007. http://doi.org/10.1111/j.1540-6261.2006.01009.x

Li, D., Nguyen, Q.N., Pham, P.K., Wei, S.X., 2011. Large foreign ownership and firm-level stock return volatility in emerging markets. Journal of Financial and Quantitative Analysis. 46, 1127-1155. http://doi.org/10.1017/s0022109011000202

Lin, C., Ma, Y., Malatesta, P., Xuan, Y., 2013. Corporate ownership structure and the choice between bank debt and public debt. Journal of Financial Economics. 109, 517-534. http://doi.org/10.1016/j.jfineco.2013.03.006

Luhmann, N., 1979. Trust and power. Wiley: Chichester, UK

Lui, D., Markov, S., Tamayo, A.N.E., 2007. What makes a stock risky? Evidence from sellside analysts' risk ratings. Journal of Accounting Research. 45, 629-665. http://doi.org/10.1111/j.1475-679X.2007.00242.x

Mandelker, G.N., Rhee, S.G., 1984. The impact of the degrees of operating and financial leverage on systematic risk of common stock. Journal of Financial and Quantitative Analysis. 19, 45-57. http://doi.org/10.2307/2331000

Mayer, R.C., Davis, J.H., Schoorman, F.D., 1995. An integrative model of organizational trust. Academy of Management Review. 20, 709-734. http://doi.org/10.2307/258792

Medrano, J.D., 2011. World map of interpersonal trust. http://www.jdsurvey.net/jds/jdsurveyMaps.jsp?Idioma=I&SeccionTexto=04&NOID=104, Accessed 10 October, 2020. Michalos, A.C., 1990. The impact of trust on business, international security and the quality of life. Journal of Business Ethics. 9, 619-638. http://doi.org/10.1007/bf00383390

Miller, K.D., Bromiley, P., 1990. Strategic risk and corporate performance: An analysis of alternative risk measures. The Academy of Management Journal. 33, 756-779. http://doi.org/10.5465/256289

Mishra, D.R., 2011. Multiple large shareholders and corporate risk taking: Evidence from East Asia. Corporate Governance: An International Review. 19, 507-528. http://doi.org/10.1111/j.1467-8683.2011.00862.x

Mitton, T., 2006. Stock market liberalization and operating performance at the firm level. Journal of Financial Economics. 81, 625-647. http://doi.org/10.1016/j.jfineco.2005.09.001

Morck, R., Yeung, B., Yu, W., 2000. The information content of stock markets: Why do emerging markets have synchronous stock price movements? Journal of Financial Economics. 58, 215-260. http://doi.org/10.1016/S0304-405X(00)00071-4

Nooteboom, B., 1996. Trust, opportunism and governance: A process and control model. Organization Studies. 17, 985-1010. http://doi.org/10.1177/017084069601700605

North, D.C., 1994. Economic performance through time. The American Economic Review. 84, 359-368

Paligorova, T. 2010. Corporate risk taking and ownership structure. Bank of Canada, Staff Working Papers 10-3. http://www.bankofcanada.ca/wp-content/uploads/2010/05/wp10-3.pdf

Pastor, L., Veronesi, P., 2003. Stock valuation and learning about profitability. The Journal of Finance. 58, 1749-1789. http://doi.org/10.1111/1540-6261.00587

Paxton, P., 2002. Social capital and democracy: An interdependent relationship. American Sociological Review. 67, 254-277. http://doi.org/10.2307/3088895

Pevzner, M., Xie, F., Xin, X., 2015. When firms talk, do investors listen? The role of trust in stock market reactions to corporate earnings announcements. Journal of Financial Economics. 117, 190-223. http://doi.org/10.1016/j.jfineco.2013.08.004

Pistor, K., Raiser, M., Gelfer, S., 2000. Law and finance in transition economies. The Economics of Transition. 8, 325-368. http://doi.org/10.1111/1468-0351.00047

Pontiff, J., 2006. Costly arbitrage and the myth of idiosyncratic risk. Journal of Accounting and Economics. 42, 35-52. http://doi.org/10.1016/j.jacceco.2006.04.002

Qian, X., Cao, T., Cao, C., 2018. Institutional environment and bank loans: Evidence from 25 developing countries. Corporate Governance: An International Review. 26, 84-96. http://doi.org/10.1111/corg.12197

Rajgopal, S., Venkatachalam, M., 2011. Financial reporting quality and idiosyncratic return volatility. Journal of Accounting and Economics. 51, 1-20. http://doi.org/10.1016/j.jacceco.2010.06.001

Sabel, C.F., 1993. Studied trust: Building new forms of cooperation in a volatile economy. Human Relations. 46, 1133-1170. http://doi.org/10.1177/001872679304600907

Sapienza, P., Zingales, L., 2012. A trust crisis. International Review of Finance. 12, 123-131. http://doi.org/10.1111/j.1468-2443.2012.01152.x

Schoorman, F.D., Mayer, R.C., Davis, J.H., 2007. An integrative model of organizational trust: Past, present, and future. Academy of Management Review. 32, 344-354. http://doi.org/10.5465/amr.2007.24348410

Sila, V., Gonzalez, A., Hagendorff, J., 2016. Women on board: Does boardroom gender diversity affect firm risk? Journal of Corporate Finance. 36, 26-53. http://doi.org/10.1016/j.jcorpfin.2015.10.003

Spolaore, E., Wacziarg, R., 2009. The Diffusion of Development. The Quarterly Journal of Economics. 124, 469-529. http://doi.org/10.1162/qjec.2009.124.2.469

Staiger, D., Stock, J.H., 1997. Instrumental variables regression with weak instruments. Econometrica. 65, 557-586. http://doi.org/10.2307/2171753

Taub, B., 2018. Inconspicuousness and obfuscation: How large shareholders dynamically manipulate output and information for trading purposes. Annals of Finance. 14, 429-464. http://doi.org/10.1007/s10436-018-0334-3

Umutlu, M., Akdeniz, L., Altay-Salih, A., 2010. The degree of financial liberalization and aggregated stock-return volatility in emerging markets. Journal of Banking & Finance. 34, 509-521. http://doi.org/10.1016/j.jbankfin.2009.08.010

Vo, X.V., 2015. Foreign ownership and stock return volatility – Evidence from Vietnam. Journal of Multinational Financial Management. 30, 101-109. http://doi.org/10.1016/j.mulfin.2015.03.004

Wang, J., 2007. Foreign equity trading and emerging market volatility: Evidence from Indonesia and Thailand. Journal of Development Economics. 84, 798-811. http://doi.org/10.1016/j.jdeveco.2006.05.001

Williamson, O.E., 1993. Calculativeness, trust, and economic organization. The Journal of Law and Economics. 36, 453-486. http://doi.org/10.1086/467284

—, 2000. The new institutional economics: Taking stock, looking ahead. Journal of Economic Literature. 38, 595-613. http://doi.org/10.1257/jel.38.3.595

Xu, Y., Malkiel, B.G., 2003. Investigating the behavior of idiosyncratic volatility. The Journal of Business. 76, 613-645. http://doi.org/10.1086/377033

Zak, P.J., Knack, S., 2001. Trust and growth. The Economic Journal. 111, 295-321. http://doi.org/10.1111/1468-0297.00609

Zou, H., Adams, M.B., 2008. Corporate ownership, equity risk and returns in the People's Republic of China. Journal of International Business Studies. 39, 1149-1168. http://doi.org/10.1057/palgrave.jibs.8400394

Appendix A - Variable Definitions

Variable	Definition									
Panal A. Diale Ma	Definition									
TDICK	<u>asures</u> Square root of 250 times the weekly return standard deviation (Data source)									
INISK	Square 1001 of 250 times the weekly feturit standard deviation (Data source.									
IDISK	DataSucally. Square root of 250 times the standard deviation of the residuals from the market model									
INISIX	regression sugmented with Fama French return factors (Data source: DataStream)									
Panal R. Main ind	lanandant variabla									
SHTRUST	Represents the level of trust in the country of origin of the foreign shareholder, where									
SITIKUST	the level of trust is measured by the percentage of respondents that most people can be									
	trusted (Data source: WVS)									
TRUSTDIFF	Represents difference between the level of trust in the country of origin of the foreign									
IROSIDIT	shareholder and the country of corporate headquarters, where the level of trust is									
	measured by the percentage of respondents to the question that "most people can be									
	trusted" (Data source: WVS)									
Panel C• Firm fur	idamentals									
ROA	Return on assets, measured as the ratio of income before extraordinary items over total									
	assets (Data source: DataStream).									
ln(ASSETS)	Natural logarithm of total assets (Data source: DataStream).									
LEVERAGE	Leverage ratio, measured as total debt over total assets (Data source: DataStream).									
MB	Market-to-book value of equity (Data source: DataStream).									
ln(AGE)	Firm age, measured as the natural logarithm of 1 plus the number of years since									
	incorporation (Data source: DataStream).									
BIG4	Binary indicator that equals 1 for the existence of a reputable auditor, and 0 otherwise.									
	Reputable auditors are considered the big four, named as PwC, Deloitte and Touché,									
	Ernst and Young and KPMG (Data source: Thomson Reuters Eikon).									
D: Country contro	<u>ols</u>									
GDPGR	Annual growth rate of GDP (Data source World Bank).									
INVPROT	Sum of the anti-self-dealing index from Djankov et al. (2008) and the rule of law index									
	from Kaufmann and Kraay (2017), after standardize both indices to be between 0 and 1.									
Panel E: Instrume	ental variables									
SHETHFR	Is the ethnic fractionalization data as described in Alesina et al. (2003).									
ETHFRDIFF	Represents difference between the ethnic fractionalization in the country of origin of the									
	foreign shareholder and the country of corporate headquarters. Ethnic fractionalization									
	data are described in Alesina et al. (2003).									
Panel F: Addition	al variables for sensitivity analyses									
SHCHANGE	Binary indicator that equals 1 if the controlling foreign shareholder has changed as									
	compared to previous year, and 0 otherwise (Data source: Thomson Reuters Eikon)									

Appendix B – First Stage Results

This table reports the first-stage results of the 2SLS estimates (Columns 3 and 4) presented in Table 4. Foreign shareholder trust (SHTRUST) is regressed on all Table 4 covariates and the ethnic fractionalization of the country of origin of the foreign shareholder (SHETHFR). The z-statistics in parentheses are based on heteroskedasticity corrected robust standard errors, clustered on firms. The continuous variables are winsorized at the 1st and 99th percentiles. The statistical significance at the 10%, 5%, and 1% levels are indicated by *, **, and ***. All variables are defined in the Appendix A.

	(1)	(2)
Dependent Variables:	SHTRUST	SHTRUST
SHETHFR	-0.094***	-0.094***
	(-7.73)	(-7.73)
ROA	-0.003**	-0.003**
	(-2.11)	(-2.11)
ln(ASSETS)	-0.001	-0.001
	(-1.23)	(-1.23)
LEVERAGE	-0.001	-0.001
	(-0.87)	(-0.87)
MB	0.001*	0.001*
	(1.95)	(1.95)
ln(AGE)	0.005**	0.005**
	(2.39)	(2.39)
BIG4	0.002	0.002
	(0.33)	(0.33)
GDPGR	0.001**	0.001**
	(2.05)	(2.05)
INVPROT	0.058***	0.058***
	(5.27)	(5.27)
(intercept)	0.222***	0.222***
	(4.97)	(4.97)
Year & Industry dummies	Yes	Yes
Adj. R^2	0.094	0.094
F-Statistic	59.79	59.79
Partial R^2	0.0266	0.0266
Observations	12,496	12,496

Table 1 Sample selection process

Sample selection stages	Number of firms	Number of firm years
Companies with common support between Eikon and DataStream (2000-2016) controlled by a foreign shareholder, for which data for the country of origin of the foreign shareholder are available through the World Values Survey.	5,243	19,065
Deleted: Companies with missing stock price information data from DataStream.	337	1,153
Deleted: Companies with missing financial data for our empirical model.	871	4,138
<u>Deleted:</u> Observations of companies don't meet the four companies per country and two observations per company criterion.	1,240	1,278
Final sample.	2,795	12,496

Table 2 Country distribution of observations

This table presents the country distribution of unique firms included in our sample, accompanied with the relevant number
of observations. TRUST denotes the level of trust in the country of corporate headquarters and SHTRUST denotes the level
of trust in the country of origin of the foreign shareholder, where the level of trust is measured by the percentage of
respondents that most people can be trusted (based on the WVS).

No	Country	Firms	Obs	%	TRUST	SHTRUST
1	Argentina	7	44	0.35	0.185	0.227
2	Australia	297	1,185	9.48	0.494	0.349
3	Austria	5	16	0.13	0.339	0.374
4	Belgium	21	72	0.58	0.307	0.400
5	Brazil	49	242	1.94	0.077	0.351
6	Canada	281	1,088	8.71	0.412	0.361
7	Chile	21	157	1.26	0.149	0.273
8	China	160	770	6.16	0.566	0.358
9	Colombia	6	37	0.30	0.060	0.314
10	Denmark	7	20	0.16	0.665	0.336
11	Egypt	12	63	0.50	0.213	0.302
12	Finland	6	16	0.13	0.580	0.423
13	France	57	247	1.98	0.198	0.358
14	Germany	106	561	4.49	0.399	0.364
15	Hungary	5	25	0.20	0.276	0.331
16	India	211	1,160	9.28	0.185	0.374
17	Indonesia	37	98	0.78	0.421	0.284
18	Ireland	18	55	0.44	0.358	0.348
19	Israel	6	17	0.14	0.235	0.311
20	Italy	21	84	0.67	0.295	0.369
21	Japan	306	1,434	11.48	0.371	0.363
22	Jordan	23	122	0.98	0.146	0.293
23	Luxembourg	6	17	0.14	0.259	0.338
24	Malaysia	73	381	3.05	0.085	0.397
25	Mexico	15	84	0.67	0.136	0.347
26	Netherlands	37	203	1.62	0.566	0.330
27	New Zealand	21	98	0.78	0.522	0.422
28	Norway	11	28	0.22	0.737	0.419
29	Pakistan	5	12	0.10	0.236	0.304
30	Peru	19	135	1.08	0.077	0.265
31	Philippines	16	61	0.49	0.041	0.416
32	Poland	63	372	2.98	0.207	0.345
33	Russia	42	188	1.50	0.275	0.118
34	Singapore	78	355	2.84	0.350	0.319
35	South Africa	15	65	0.52	0.188	0.389
36	South Korea	104	519	4.15	0.272	0.379
37	Spain	20	88	0.70	0.215	0.312
38	Sweden	7	16	0.13	0.664	0.419
39	Switzerland	38	137	1.10	0.512	0.333
40	Thailand	66	370	2.96	0.343	0.351
41	Turkey	18	51	0.41	0.102	0.388
42	United Kingdom	127	479	3.83	0.299	0.375
43	United States of America	352	1,324	10.60	0.371	0.398
	Total	2,795	12,496	100	-	-

Table 3 Descriptive statistics (N = 12,496)

Table 5 Descriptive statistics ($N = 12,496$)								
This table presents the descriptive statistics of the variables employed in our analyses. The continuous variables are								
winsorized at the 1st and 99th percentiles. All variables are defined in the Appendix A.								

Variable	25th	Mean	Median	75th	StDev
TRISK	0.629	1.254	0.957	1.523	0.987
IRISK	0.544	1.153	0.843	1.397	0.97
SHTRUST	0.3	0.357	0.359	0.391	0.121
ROA	-0.049	-0.199	0.033	0.081	1.172
ln(ASSETS)	11.07	13.904	14.017	16.686	4.007
LEVERAGE	0	0.576	0.179	0.69	1.548
MB	0.68	1.907	1.3	2.42	6.829
ln(AGE)	2.197	2.865	2.996	3.689	1.108
BIG4	0	0.581	1	1	0.493
GDPGR	1.654	3.221	2.766	4.694	3.263
INVPROT	1.156	1.371	1.375	1.598	0.311

Table 4 Foreign shareholder trust and firm risk

This table reports the OLS estimates (Columns 1 and 2) and the 2SLS estimates (Columns 3 and 4) of the effect of foreign shareholder trust on firm market risk. The dependent variables in Columns 1 and 3 is the standard deviation of each firm's weekly stock returns in each year, multiplied by the square root of 250 (TRISK). The dependent variable in Columns 2 and 4 is the standard deviation of the residuals from the market model regression augmented with Fama-French return factors, multiplied by the square root of 250 (IRISK). The key independent variable is SHTRUST, which represents the level of trust in the country of origin of the foreign shareholder, where the level of trust is measured by the percentage of respondents that most people can be trusted. We instrument foreign shareholder trust (Columns 3 and 4) with SHETHFR, representing the ethnic fractionalization of the country of origin of the foreign shareholder trust (Columns 1 and 2, and 3 and 4, in parentheses are based on heteroskedasticity corrected robust standard errors, clustered on firms. The continuous variables are winsorized at the 1st and 99th percentiles. The statistical significance at the 10%, 5%, and 1% levels are indicated by *, **, and ***. All variables are defined in the Appendix A.

	(1)	(2)	(3)	(4)	
Statistical Method:	OLS E	stimates	2SLS Estimates		
Dependent Variables:	TRISK	IRISK	TRISK	IRISK	
SHTRUST	-0.312***	-0.298***	-1.985***	-1.707***	
	(-3.28)	(-3.22)	(-3.03)	(-2.73)	
ROA	-0.103***	-0.097***	-0.108***	-0.101***	
	(-4.91)	(-4.74)	(-5.02)	(-4.85)	
ln(ASSETS)	-0.089***	-0.096***	-0.091***	-0.098***	
	(-19.65)	(-21.92)	(-18.23)	(-20.52)	
LEVERAGE	0.009	0.004	0.007	0.003	
	(1.26)	(0.68)	(0.91)	(0.41)	
MB	-0.003	-0.003	-0.002	-0.002	
	(-1.53)	(-1.49)	(-1.09)	(-1.10)	
ln(AGE)	-0.072***	-0.065***	-0.060***	-0.055***	
	(-6.20)	(-5.78)	(-4.60)	(-4.41)	
BIG4	-0.308***	-0.317***	-0.310***	-0.319***	
	(-14.06)	(-14.86)	(-12.66)	(-13.61)	
GDPGR	0.004	0.003	0.007	0.006	
	(0.99)	(0.82)	(1.57)	(1.37)	
INVPROT	0.070*	0.072*	0.193***	0.179***	
	(1.68)	(1.80)	(3.32)	(3.30)	
(intercept)	2.874***	2.941***	3.347***	3.354***	
	(14.44)	(15.26)	(12.53)	(13.20)	
Year & Industry dummies	Yes	Yes	Yes	Yes	
Hausman test			43.473***	35.24***	
Adj. R^2	0.423	0.439			
Mean VIF	1.313	1.327	1.313	1.327	
Observations	12,496	12,496	12,496	12,496	
First stage	(1)	(2)	(3)	(4)	
SHETHFR			-0.094***	-0.094***	
			(-7.73)	(-7.73)	
<i>F</i> -Statistic			59.79	59.79	
Partial R^2			0.0266	0.0266	
Control Variables			Yes	Yes	
Year & Industry dummies	•	•	Yes	Yes	

Table 5 Variations in investor protection, foreign shareholder trust and firm risk

This table reports the 2SLS analysis results of the effect of foreign shareholder trust on firm market risk, after separating the sample, using the sample median of INVPROT in each year, into countries with low (high) investor protections in Columns 1 and 2 (3 and 4). The dependent variable in Columns 1 and 3 is the standard deviation of each firm's weekly stock returns in each year, multiplied by the square root of 250 (TRISK). The dependent variable in Columns 2 and 4 is the standard deviation of the residuals from the market model regression augmented with Fama-French return factors, multiplied by the square root of 250 (IRISK). The key independent variable is SHTRUST, which represents the level of trust in the country of origin of the foreign shareholder, where the level of trust is measured by the percentage of respondents that most people can be trusted. We instrument foreign shareholder. The *z*-statistics in parentheses are based on heteroskedasticity corrected robust standard errors, clustered on firms. The continuous variables are winsorized at the 1st and 99th percentiles. The statistical significance at the 10%, 5%, and 1% levels are indicated by *, **, and ***. All variables are defined in the Appendix A.

	(1)	(2)	(3)	(4)
	Low Invest	or Protection	High Investor	r Protection
Dependent Variables:	TRISK	IRISK	TRISK	IRISK
SHTRUST	-4.119***	-3.340**	-0.953	-0.958
	(-2.64)	(-2.43)	(-1.37)	(-1.41)
ROA	-0.186***	-0.177***	-0.048**	-0.042*
	(-3.67)	(-3.67)	(-2.02)	(-1.83)
ln(ASSETS)	-0.058***	-0.064***	-0.131***	-0.138***
	(-6.99)	(-8.48)	(-19.68)	(-21.36)
LEVERAGE	0.003	-0.001	0.009	0.006
	(0.24)	(-0.09)	(0.81)	(0.63)
MB	0.001	0.001	-0.004*	-0.004
	(0.35)	(0.22)	(-1.69)	(-1.64)
ln(AGE)	-0.037	-0.038	-0.054***	-0.050***
	(-1.19)	(-1.36)	(-3.51)	(-3.33)
BIG4	-0.126***	-0.147***	-0.321***	-0.326***
	(-3.41)	(-4.50)	(-9.07)	(-9.42)
GDPGR	0.054***	0.049***	-0.052***	-0.047***
	(5.50)	(5.56)	(-6.08)	(-5.72)
(intercept)	3.364***	3.236***	4.148***	4.188***
	(6.87)	(7.52)	(11.67)	(12.02)
Year & Industry dummies	Yes	Yes	Yes	Yes
Hausman test	39.273***	27.585***	10.478***	11.064***
Mean VIF	1.193	1.202	1.414	1.430
Observations	5,849	5,849	6,647	6,647
First stage	(1)	(2)	(3)	(4)
SHETHFR	-0.063***	-0.063***	-0.118***	-0.118***
	(-3.41)	(-3.41)	(-7.31)	(-7.31)
F-Statistic	11.64	11.64	53.42	53.42
Partial R ²	0.0107	0.0107	0.045	0.045
Control Variables	Yes	Yes	Yes	Yes
Year & Industry dummies	Yes	Yes	Yes	Yes

Table 6 Variations in institutional context strength, foreign shareholder trust and firm risk

This table reports the 2SLS analysis results of the effect of foreign shareholder trust on firm market risk, after separating the sample into strong (weak) institutional contexts in Columns 1 and 2 (3 and 4). The dependent variable in Columns 1 and 3 is the standard deviation of each firm's weekly stock returns in each year, multiplied by the square root of 250 (TRISK). The dependent variable in Columns 2 and 4 is the standard deviation of the residuals from the market model regression augmented with Fama-French return factors, multiplied by the square root of 250 (IRISK). The key independent variable is SHTRUST, which represents the level of trust in the country of origin of the foreign shareholder, where the level of trust is measured by the percentage of respondents that most people can be trusted. We instrument foreign shareholder trust with SHETHFR, representing the ethnic fractionalization of the country of origin of the foreign shareholder. The *z*-statistics in parentheses are based on heteroskedasticity corrected robust standard errors, clustered on firms. The continuous variables are winsorized at the 1st and 99th percentiles. The statistical significance at the 10%, 5%, and 1% levels are indicated by *, **, and ***. All variables are defined in the Appendix A.

	(1)	(2)	(3) (4)			
	Strong Institu	tional Context	Weak Instituti	onal Context		
Dependent Variables:	TRISK	IRISK	TRISK	IRISK		
SHTRUST	-2.490	-2.229	-1.299***	-1.142***		
	(-1.33)	(-1.24)	(-3.27)	(-3.00)		
ROA	-0.085***	-0.078***	-0.133**	-0.124**		
	(-3.74)	(-3.56)	(-2.21)	(-2.12)		
ln(ASSETS)	-0.111***	-0.119***	-0.069***	-0.075***		
	(-15.17)	(-16.93)	(-8.33)	(-9.36)		
LEVERAGE	-0.000	-0.001	0.018*	0.010		
	(-0.04)	(-0.12)	(1.69)	(1.03)		
MB	-0.002	-0.002	-0.002	-0.002		
	(-0.72)	(-0.72)	(-0.48)	(-0.63)		
ln(AGE)	-0.050***	-0.044***	-0.106***	-0.102***		
	(-3.49)	(-3.18)	(-3.69)	(-3.74)		
BIG4	-0.311***	-0.328***	-0.197***	-0.193***		
	(-7.84)	(-8.58)	(-6.63)	(-6.73)		
GDPGR	-0.049***	-0.046***	0.030***	0.024***		
	(-4.22)	(-4.17)	(4.96)	(4.13)		
INVPROT	0.459**	0.426**	-0.043	-0.032		
	(2.20)	(2.14)	(-0.65)	(-0.51)		
(intercept)	3.548***	3.590***	3.072***	3.100***		
	(8.48)	(8.95)	(10.56)	(11.41)		
Year & Industry dummies	Yes	Yes	Yes	Yes		
Hausman test	7.733***	6.708***	41.242***	34.229***		
Mean VIF	1.383	1.398	1.205	1.211		
Observations	7,753	7,753	4,743	4,743		
First stage	(1)	(2)	(3)	(4)		
SHETHFR	-0.045***	-0.045***	-0.204***	-0.204***		
	(-3.00)	(-3.00)	(-9.80)	(-9.80)		
F-Statistic	10.01	10.01	96.11	96.11		
Partial R^2	0.006	0.006	0.1199	0.1199		
Control Variables	Yes	Yes	Yes	Yes		
Year & Industry dummies	Yes	Yes	Yes	Yes		

Table 7 Variations in investors' horizon, foreign shareholder trust and firm risk

This table reports the 2SLS analysis results of the effect of foreign shareholder trust on firm market risk, after separating the sample as per foreign major shareholders investment horizon, namely long-term (Columns 1 and 2) and short-term (Columns 3 and 4). Columns 5 and 6 report the t-statistics for Wald tests used to compare the difference in coefficients between regression results, respectively, of Columns 1 and 3 and Columns 2 and 4. The dependent variable in Columns 1 and 3 is the standard deviation of each firm's weekly stock returns in each year, multiplied by the square root of 250 (TRISK). The dependent variable in Columns 2 and 4 is the standard deviation of the residuals from the market model regression augmented with Fama-French return factors, multiplied by the square root of 250 (IRISK). The key independent variable is SHTRUST, which represents the level of trust in the country of origin of the foreign shareholder, where the level of trust is measured by the percentage of respondents that most people can be trusted. We instrument foreign shareholder. The z-statistics in parentheses are based on heteroskedasticity corrected robust standard errors, clustered on firms. The continuous variables are winsorized at the 1st and 99th percentiles. The statistical significance at the 10%, 5%, and 1% levels are indicated by *, **, and ***. All variables are defined in the Appendix A.

	(1)	(2)	(2) (3)			
	Long Term In	vestor Horizon	Short Term Inve	estor Horizon		
Dependent Variables:	TRISK	IRISK	TRISK	IRISK		
SHTRUST	-1.251***	-1.027***	-35.114	-31.818		
	(-3.01)	(-2.83)	(-0.08)	(-0.08)		
ROA	-0.091***	-0.085***	-0.204	-0.189		
	(-3.78)	(-3.61)	(-0.15)	(-0.15)		
ln(ASSETS)	-0.087***	-0.095***	-0.235	-0.232		
	(-17.02)	(-19.34)	(-0.16)	(-0.17)		
LEVERAGE	0.010	0.005	-0.112	-0.106		
	(1.27)	(0.70)	(-0.08)	(-0.09)		
MB	-0.003	-0.003	0.007	0.006		
	(-1.46)	(-1.48)	(0.10)	(0.09)		
ln(AGE)	-0.058***	-0.053***	-0.194	-0.182		
	(-4.42)	(-4.17)	(-0.16)	(-0.16)		
BIG4	-0.287***	-0.296***	-0.384	-0.390		
	(-11.65)	(-12.48)	(-0.71)	(-0.80)		
GDPGR	0.001	0.000	0.033	0.031		
	(0.31)	(0.07)	(0.09)	(0.09)		
INVPROT	0.154***	0.145***	4.507	4.146		
	(2.99)	(2.99)	(0.08)	(0.08)		
(intercept)	2.917***	2.894***	11.496	10.699		
	(11.05)	(11.41)	(0.11)	(0.11)		
Year & Industry dummies	Yes	Yes	Yes	Yes		
Hausman test	25.893***	19.862***	0.174	0.149		
Mean VIF	1.299	1.314	1.455	1.465		
Observations	9,894	9,894	439	439		
First stage	(1)	(2)	(3)	(4)		
SHETHFR	-1.251**	-1.251**	-35.114	-35.114		
	(-2.22)	(-2.22)	(-0.08)	(-0.08)		
F-Statistic	68.45	68.45	0.01	0.01		
Partial R^2	0.0363	0.0363	0	0		
Control Variables	Yes	Yes	Yes	Yes		
Year & Industry dummies	Yes	Yes	Yes	Yes		

Table 8 Robustness of main results and extensions, using as main independent variable the difference between foreign shareholder trust and local trust

This table reports the 2SLS analysis results of the effect of the difference between the level of trust in the country of origin of the foreign shareholder and the country of corporate headquarters, on firm market risk. We replicate our main results in Columns 1 and 2, and split the sample into: a) low (high) investor protection in Columns 3 and 4 (5 and 6); b) strong (weak) institutional contexts in Columns 7 and 8 (9 and 10); and low (high) monitoring from institutional investors in Columns 11 and 12 (13 and 14). The dependent variable in Columns 1, 3, 5, 7, 9, 11, and 13 is the standard deviation of each firm's weekly stock returns in each year, multiplied by the square root of 250 (TRISK). The dependent variable in Columns 2, 4, 6, 8, 10, 12, and 14 is the standard deviation of the residuals from the market model regression augmented with Fama-French return factors, multiplied by the square root of 250 (IRISK). The key independent variable is TRUSTDIFF, and represents the difference between the level of trust in the country of origin of the foreign shareholder and the country of origin of the foreign shareholder and the country of origin of the foreign shareholder and the country of origin of the foreign shareholder and the country of origin of the foreign shareholder and the country of origin of the foreign shareholder and the country of origin of the foreign shareholder and the country of origin of the foreign shareholder and the country of origin of the foreign shareholder and the country of origin of the foreign shareholder and the country of origin of the foreign shareholder and the country of origin of the foreign shareholder and the country of origin of the foreign shareholder and the country of origin of the foreign shareholder and the country of origin of the foreign shareholder and the country of origin of the foreign shareholder and the country of origin of the foreign shareholder and the country of origin of the foreign shareholder and the country of corporate headquarters. The *z*-statistics in

$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$		(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)	(13)	(14)
Dependent Main results Prote-tion Context Context Context Hoizon Hoizon Variables: TRISK IRISK TRISK IRISK TRISK IRISK TRISK <		Main I	Deculte	Low In	ivestor	High Ir	nvestor	Strong Ins	titutional	Weak Inst	titutional	Long Tern	n Investor	Short Terr	n Investor
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	Dependent	Iviani I	Cesuits	Prote	ction	Prote	ction	Con	text	Cont	text	Hori	zon	Hori	izon
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	Variables:	TRISK	IRISK	TRISK	IRISK	TRISK	IRISK	TRISK	IRISK	TRISK	IRISK	TRISK	IRISK	TRISK	IRISK
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	TRUSTDIFF	-0.835***	-0.755***	-1.661***	-1.418***	-0.339	-0.412	2.840	2.340	-1.985***	-1.854***	-0.544**	-0.488**	-1.621	-1.487
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		(-3.39)	(-3.16)	(-5.11)	(-4.63)	(-1.01)	(-1.26)	(1.53)	(1.32)	(-6.10)	(-5.96)	(-2.33)	(-2.16)	(-0.93)	(-0.86)
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	ROA	-0.104***	-0.098***	-0.158***	-0.154***	-0.050**	-0.045**	-0.082***	-0.076***	-0.127**	-0.120**	-0.090***	-0.084***	-0.102*	-0.097
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		(-4.96)	(-4.79)	(-2.93)	(-3.00)	(-2.15)	(-1.98)	(-3.86)	(-3.67)	(-2.13)	(-2.07)	(-3.74)	(-3.58)	(-1.70)	(-1.64)
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	ln(ASSETS)	-0.090***	-0.097***	-0.056***	-0.062***	-0.127***	-0.134***	-0.103***	-0.111***	-0.078***	-0.083***	-0.086***	-0.094***	-0.125***	-0.132***
LEVERAGE 0.004 0.001 -0.003 -0.006 0.007 0.004 0.013 0.010 0.014 0.006 0.008 0.003 -0.015 -0.017 (0.64) (0.11) (-0.25) (-0.66) (0.63) (0.42) (1.18) (1.00) (1.17) (0.59) (1.07) (0.47) (-0.39) (-0.49) MB -0.002 -0.002 0.003 0.002 -0.004 -0.004 -0.003 0.002 0.003 -0.003 -0.000 -0.000 (-0.96) (-0.96) (1.10) (0.90) (-1.67) (-1.60) (-1.33) (-1.26) (0.69) (0.52) (-1.34) (1.34) (0.07) (-0.00) In(AGE) -0.059*** -0.053*** -0.051*** -0.035** -0.031* 0.013 0.010 -0.057*** -0.015*** -0.105** -0.101** (-4.79) (-4.47) (-1.07) (-1.12) (-3.58) (-3.41) (-1.98) (-1.93) (0.33) (0.26) (-3.21** -0.424*** BIG4 -0.333*** -0.34*** -0.256*** -0.25**		(-19.46)	(-21.67)	(-8.70)	(-10.13)	(-20.25)	(-21.96)	(-15.53)	(-18.14)	(-8.93)	(-9.84)	(-17.64)	(-19.87)	(-6.37)	(-6.93)
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	LEVERAGE	0.004	0.001	-0.003	-0.006	0.007	0.004	0.013	0.010	0.014	0.006	0.008	0.003	-0.015	-0.017
MB -0.002 -0.002 0.003 0.002 $-0.004*$ -0.004 -0.003 0.002 0.002 -0.003 -0.003 -0.003 0.000 -0.003 $-0.051***$ $-0.051***$ $-0.101**$ In(AGE) $-0.333***$ -0.025 -0.025 $-0.031**$ $-0.328***$ $-0.257***$ $-0.328***$ $-0.269***$ $-0.302***$ $-0.310***$ $-0.421***$ $-0.424***$ In(4.43) (-15.10) (-7.18) (-7.66) (-9.53) (-9.99) (-7.15) (-7.15) (-7.15) (-12.52) (-13.21) (-4.78) (-5.03) GDPGR 0.006 0.006 0.005 <td></td> <td>(0.64)</td> <td>(0.11)</td> <td>(-0.25)</td> <td>(-0.66)</td> <td>(0.63)</td> <td>(0.42)</td> <td>(1.18)</td> <td>(1.00)</td> <td>(1.17)</td> <td>(0.59)</td> <td>(1.07)</td> <td>(0.47)</td> <td>(-0.39)</td> <td>(-0.49)</td>		(0.64)	(0.11)	(-0.25)	(-0.66)	(0.63)	(0.42)	(1.18)	(1.00)	(1.17)	(0.59)	(1.07)	(0.47)	(-0.39)	(-0.49)
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	MB	-0.002	-0.002	0.003	0.002	-0.004*	-0.004	-0.004	-0.003	0.002	0.002	-0.003	-0.003	0.000	-0.000
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		(-0.96)	(-0.96)	(1.10)	(0.90)	(-1.67)	(-1.60)	(-1.33)	(-1.26)	(0.69)	(0.52)	(-1.34)	(-1.34)	(0.07)	(-0.00)
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	ln(AGE)	-0.059***	-0.053***	-0.025	-0.025	-0.054***	-0.051***	-0.035**	-0.031*	0.013	0.010	-0.057***	-0.051***	-0.105**	-0.101**
BIG4 -0.333*** -0.340*** -0.256*** -0.257*** -0.30*** -0.310*** -0.277*** -0.269*** -0.302*** -0.310*** -0.421*** -0.424*** (-14.34) (-15.10) (-7.18) (-7.66) (-9.53) (-9.99) (-8.34) (-9.53) (-7.15) (-7.25) (-12.52) (-13.21) (-4.78) (-5.03) GDPGR 0.006 0.005 0.034*** 0.032*** -0.044*** -0.039*** -0.026** -0.004 -0.008 0.000 -0.001 -0.005 -0.004 (1.33) (1.15) (6.82) (6.87) (-4.92) (-4.43) (-2.32) (-2.55) (-0.60) (-1.11) (0.07) (-0.13) (-0.26) (-0.20) INVPROT 0.020 0.028 0.251*** 0.236*** -0.129* -0.110 0.049 0.055 -0.117 -0.044 (0.41) (0.59) 2.223*** 3.779*** 3.809*** 3.370*** 3.418*** 3.090*** 3.134*** 2.663*** 2.688*** 2.678*** 2.705***		(-4.79)	(-4.47)	(-1.07)	(-1.12)	(-3.58)	(-3.41)	(-1.98)	(-1.93)	(0.33)	(0.26)	(-4.42)	(-4.06)	(-2.30)	(-2.27)
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	BIG4	-0.333***	-0.340***	-0.256***	-0.257***	-0.330***	-0.337***	-0.310***	-0.328***	-0.277***	-0.269***	-0.302***	-0.310***	-0.421***	-0.424***
GDPGR 0.006 0.005 0.034*** 0.032*** -0.044*** -0.039*** -0.025** -0.004 -0.008 0.000 -0.001 -0.005 -0.004 (1.33) (1.15) (6.82) (6.87) (-4.92) (-4.43) (-2.32) (-2.55) (-0.60) (-1.11) (0.07) (-0.13) (-0.26) (-0.20) INVPROT 0.020 0.028		(-14.34)	(-15.10)	(-7.18)	(-7.66)	(-9.53)	(-9.99)	(-8.34)	(-9.53)	(-7.15)	(-7.25)	(-12.52)	(-13.21)	(-4.78)	(-5.03)
(1.33) (1.15) (6.82) (6.87) (-4.92) (-4.43) (-2.32) (-2.55) (-0.60) (-1.11) (0.07) (-0.13) (-0.26) (-0.20) INVPROT 0.020 0.028 0.251*** 0.236*** -0.129* -0.110 0.049 0.055 -0.117 -0.044 (0.41) (0.59) (2.99) (3.12) (-1.75) (-1.57) (0.99) (1.17) (-0.43) (-0.16) (intercept) 2.977*** 3.037*** 2.124*** 2.223*** 3.779*** 3.809*** 3.370*** 3.418*** 3.090*** 3.134*** 2.663*** 2.688*** 2.678*** 2.705***	GDPGR	0.006	0.005	0.034***	0.032***	-0.044***	-0.039***	-0.025**	-0.026**	-0.004	-0.008	0.000	-0.001	-0.005	-0.004
INVPROT 0.020 0.028 0.251*** 0.236*** -0.129* -0.110 0.049 0.055 -0.117 -0.044 (0.41) (0.59) (2.99) (3.12) (-1.75) (-1.57) (0.99) (1.17) (-0.43) (-0.16) (intercept) 2.977*** 3.037*** 2.124*** 2.223*** 3.779*** 3.809*** 3.370*** 3.418*** 3.090*** 3.134*** 2.663*** 2.688*** 2.678*** 2.705***		(1.33)	(1.15)	(6.82)	(6.87)	(-4.92)	(-4.43)	(-2.32)	(-2.55)	(-0.60)	(-1.11)	(0.07)	(-0.13)	(-0.26)	(-0.20)
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	INVPROT	0.020	0.028					0.251***	0.236***	-0.129*	-0.110	0.049	0.055	-0.117	-0.044
(intercept) 2.977*** 3.037*** 2.124*** 2.223*** 3.779*** 3.809*** 3.370*** 3.418*** 3.090*** 3.134*** 2.663*** 2.688*** 2.678*** 2.705***		(0.41)	(0.59)					(2.99)	(3.12)	(-1.75)	(-1.57)	(0.99)	(1.17)	(-0.43)	(-0.16)
	(intercept)	2.977***	3.037***	2.124***	2.223***	3.779***	3.809***	3.370***	3.418***	3.090***	3.134***	2.663***	2.688***	2.678***	2.705***
(14.34) (15.05) (8.54) (9.83) (14.30) (14.55) (10.17) (11.46) (10.51) (11.34) (12.37) (12.78) (3.89) (4.03)		(14.34)	(15.05)	(8.54)	(9.83)	(14.30)	(14.55)	(10.17)	(11.46)	(10.51)	(11.34)	(12.37)	(12.78)	(3.89)	(4.03)
	X Z 0 T 1 4														
Year & Industry Yes	Year & Industry dummies	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes						
Hausman test 23.882*** 20.702*** 80.607*** 60.336*** 4.925** 7.386*** 11.735*** 8.263*** 108.512*** 97.305*** 10.751*** 9.503*** 0.672 0.573	Hausman test	23.882***	20.702***	80.607***	60.336***	4.925**	7.386***	11.735***	8.263***	108.512***	97.305***	10.751***	9.503***	0.672	0.573
Mean VIF 1.320 1.333 1.200 1.209 1.425 1.441 1.376 1.391 1.237 1.243 1.306 1.321 1.442 1.452	Mean VIF	1.320	1.333	1.200	1.209	1.425	1.441	1.376	1.391	1.237	1.243	1.306	1.321	1.442	1.452

(continued on next page)

Table 8 (continued)													
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)	(13)	(14)
	Main E	Doculto	Low In	vestor	High Ir	nvestor	Strong Institutional		Weak Institutional		Long Term Investor		Short Term Investor	
Dependent	Iviaiii F	Cesuits	Prote	ction	Prote	ction	Con	text	Con	text	Hori	zon	Hori	zon
Variables:	TRISK	IRISK	TRISK	IRISK	TRISK	IRISK	TRISK	IRISK	TRISK	IRISK	TRISK	IRISK	TRISK	IRISK
Observations	12,496	12,496	5,849	5,849	6,647	6,647	7,753	7,753	4,743	4,743	9,894	9,894	439	439
First stage	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)	(13)	(14)
ETHFRDIFF	-0.153***	-0.153***	-0.187***	-0.187***	-0.137***	-0.137***	-0.039***	-0.039***	-0.220***	-0.220***	-0.171***	-0.171***	-0.072**	-0.072**
	(-13.40)	(-13.40)	(-9.10)	(-9.10)	(-10.74)	(-10.74)	(-3.05)	(-3.05)	(-9.73)	(-9.73)	(-13.45)	(-13.45)	(-2.32)	(-2.32)
F-Statistic	179.48	179.48	82.84	82.84	115.27	115.27	10.01	10.01	94.72	94.72	181.03	181.03	5.40	5.40
Partial R^2	0.0741	0.0741	0.0669	0.0669	0.0899	0.0899	0.0063	0.0063	0.1076	0.1076	0.0874	0.0874	0.024	0.024
Control Variables	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes						
Year & Industry dummies	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes						

Table 9 Robustness of main results, after controlling for changes in foreign shareholders and partitioning the sample into companies where the new foreign shareholder comes from a more/less trusting environment

This table reports the 2SLS analysis results of the effect of foreign shareholder trust on firm market risk, after augmenting our model for an indicator variable signalling change in the controlling foreign shareholder as compared to the previous year (SHCHANGE, in Columns 1 and 2) and separating our sample into firms where the new controlling foreign shareholder comes from a more (less) trusting environment as compared to the previous shareholder in Columns 3 and 4 (5 and 6). The dependent variable in Columns 1, 3, and 5 is the standard deviation of each firm's weekly stock returns in each year, multiplied by the square root of 250 (TRISK). The dependent variable in Columns 2, 4, and 6 is the standard deviation of the residuals from the market model regression augmented with Fama-French return factors, multiplied by the square root of 250 (TRISK). The key independent variable is SHTRUST, which represents the level of trust in the country of origin of the foreign shareholder, where the level of trust is measured by the percentage of respondents that most people can be trusted. We instrument foreign shareholder trust with SHETHFR, representing the ethnic fractionalization of the country of origin of the foreign shareholder. The *z*-statistics in parentheses are based on heteroskedasticity corrected robust standard errors, clustered on firms. The continuous variables are winsorized at the 1st and 99th percentiles. The statistical significance at the 10%, 5%, and 1% levels are indicated by *, **, and ***. All variables are defined in the Appendix A.

	(1)	(2)	(3)	(4)	(5)	(6)
Dependent Variables:	TRISK	IRISK	TRISK	IRISK	TRISK	IRISK
SHTRUST	-1.837***	-1.582***	-3.759***	-3.532***	1.638	1.281
	(-2.90)	(-2.61)	(-3.16)	(-3.06)	(1.21)	(0.98)
SHCHANGE	0.115***	0.107***				
	(6.20)	(5.93)				
ROA	-0.097***	-0.091***	-0.088**	-0.083**	-0.182***	-0.176***
	(-4.41)	(-4.23)	(-2.48)	(-2.39)	(-3.81)	(-3.84)
SIZE	-0.092***	-0.100***	-0.120***	-0.126***	-0.101***	-0.107***
	(-18.57)	(-20.95)	(-12.36)	(-13.36)	(-9.87)	(-10.88)
LEVERAGE	0.011	0.006	0.013	0.007	0.060**	0.058**
	(1.46)	(0.93)	(0.74)	(0.39)	(2.34)	(2.32)
MB	-0.004*	-0.004*	-0.001	-0.000	-0.020**	-0.020**
	(-1.78)	(-1.78)	(-0.15)	(-0.03)	(-2.50)	(-2.53)
LnAGE	-0.058***	-0.053***	-0.058***	-0.048**	-0.036	-0.031
	(-4.42)	(-4.17)	(-2.59)	(-2.21)	(-1.60)	(-1.42)
BIG4	-0.311***	-0.320***	-0.424***	-0.431***	-0.294***	-0.302***
	(-12.72)	(-13.64)	(-7.73)	(-8.05)	(-5.62)	(-5.97)
GDPGR	0.006	0.004	0.006	0.005	0.021	0.022*
	(1.23)	(0.99)	(0.61)	(0.54)	(1.55)	(1.70)
INVPROT	0.180***	0.170***	0.258*	0.247*	-0.011	0.020
	(3.23)	(3.24)	(1.89)	(1.89)	(-0.06)	(0.11)
(intercept)	3.090***	3.059***	4.327***	4.273***	2.541***	2.546***
	(11.58)	(11.99)	(7.79)	(7.95)	(5.05)	(5.26)
			T 7			
Year & Industry dummies	Yes	Yes	Yes	Yes	Yes	Yes
Hausman test	38.8/4***	31.68***	22.648***	21.529***	2.126*	1.5/3*
Mean VIF	1.288	1.301	1.380	1.394	1.3//	1.392
Observations	11,997	11,997	1,927	1,927	1,191	1,191
First stage	(1)	(2)	(3)	(4)	(5)	(6)
SHETHFR	-0.096***	-0.096***	-0.105***	-0.105***	0.096***	0.096***
	(-7.75)	(-7.75)	(-6.38)	(-6.38)	(5.52)	(5.52)
F-Statistic	60.09	60.09	40.69	40.69	30.44	30.44
Partial R^2	0.0277	0.0277	0.0439	0.0439	0.0374	0.0374
Control Variables	Vac	Var	Vaa	Vac	Vaa	Vac
Voor & Industry dummins	res	r es Voc	res	r es Vas	r es Vez	res
i ear & industry dummies	res	res	res	res	res	res

Supplementary material for:

Major Shareholders' Trust and Market Risk: Substituting Weak Institutions with Trust

This supplementary material provides the following empirical results:

- Table IA 1 provides the Pearson correlation matrix of the variables used in our main analyses.
- Table IA 2 presents results of the effect of foreign shareholder trust on firm market risk, after controlling for sample selection bias through incorporating the inverse Mills ratio (LABDA) alongside with the first-stage results of the Heckman correction process.
- Table IA 3 presents the effect of foreign shareholder trust on alternative specifications of firm risk, namely: a) idiosyncratic risk (IRISKMM) using a simple model market regression (without the Fama and French (1993) three factors); b) logarithmic transformation of our risk measures (ln(TRISK) and ln(IRISK)); c) the three (five) years standard deviation of ROA and ROE, namely the SD3ROA and SD3ROE (SD5ROA and SD5ROE).
- Table IA 4 presents the effect of foreign shareholder trust on firm market risk, using the following additional tests: a) alternative specifications of foreign shareholder trust (SHTRUST_INDEX through Medrano (2011) and linear interpolated values of shareholder trust SHTRUST_INTP); b) using the genetic distance between the population of the country of firm headquarters and that of the origin of foreign shareholder (GENETDIST) as an alternative instrument for SHTRUST; c) alternative samples (i.e., excluding the years 2015 and 2016 as they are not covered through WVS and excluding the crisis period).
- Table IA 5 presents the effect of foreign shareholder trust on firm market risk, using alternative thresholds of foreign blockholders, namely at 10%, 20% and 25%.

- Table IA 6 presents the effect of foreign shareholder trust on firm market risk, after considering for country fixed effects and clustering standard errors at country level.
- Table IA 7 presents the effect of foreign shareholder trust on firm market risk, after controlling for additional firm-level characteristics, such as operating leverage (OPLEV); dividend pay-out (DIVPAYOUT); research and development expenditure (RDEXP); capital expenditure (CAPEX); volatility of operating cash flows (STDCF); and product market competition (HHI).
- Table IA 8 presents the effect of foreign shareholder trust on firm market risk, after controlling for additional firm-level characteristics such as cross-listed status (CROSSLISTED) and ownership by foreign and institutional investors (FOR_OWN and INST_OWN).
- Table IA 9 presents the effect of foreign shareholder trust on firm market risk, after considering for the geographical distance (ln(GEODIST)) and share boarders (SHAREBOARDER) between the firm and the foreign shareholder, as well as after controlling for Hofstede's (2001) cultural dimensions at the country of foreign shareholder.
- Table IA 10 presents the effect of foreign shareholder trust on firm market risk, after controlling for additional country-level characteristics at the country of corporate headquarters, namely for Hofstede's (2001) cultural dimensions; stock market development (MCAPCNT); the country's legal origin (COMLAW); income inequalities (GINI); and for each country's credit rating of sovereign debt (SPRT, FITCHRT, and MOODYSRT).
- Table IA 11 presents the effect of foreign shareholder trust on firm market risk, after excluding countries with high representation in our sample, namely US, Japan, India, Canada, and Australia.

- Table IA 12 presents the effect of foreign shareholder trust on firm market risk, using alternative definitions of investor protection, namely namely the rule of law (RULAW) and the control for corruption (CORRUP) indices through the World Bank (Kaufmann and Kraay, 2017).
- Table IA 13 presents the effect of foreign shareholder trust on firm market risk, using alternative definitions of the quality of the institutional environment, namely the voice and accountability (VOICACC) and the government effectiveness (GOVEFF) indices through the World Bank (Kaufmann and Kraay, 2017).

References

Fama, E.F., French, K.R., 1993. Common risk factors in the returns on stocks and bonds. Journal of Financial Economics. 33, 3-56. http://dx.doi.org/10.1016/0304-405X(93)90023-5

Hasan, M.M., Habib, A., 2019. Social capital and idiosyncratic return volatility. Australian Journal of Management. 44, 3-31. http://doi.org/10.1177/0312896217717573

Hofstede, G.H., 2001. Culture's consequences: Comparing values, behaviors, institutions and organizations across nations. Sage Publications: Thousand Oaks, CA

[dataset] Kaufmann, D., Kraay, A., 2017. The worldwide governance indicators (WGI) project. The World Bank. http://info.worldbank.org/governance/wgi/#home

La Porta, R., Lopez-de-Silanes, F., Shleifer, A., 1999. Corporate ownership around the world. The Journal of Finance. 54, 471-517. http://doi.org/10.1111/0022-1082.00115

Medrano, J.D., 2011. World map of interpersonal trust. http://www.jdsurvey.net/jds/jdsurveyMaps.jsp?Idioma=I&SeccionTexto=04&NOID=104, Accessed 10 January, 2020.

Sila, V., Gonzalez, A., Hagendorff, J., 2016. Women on board: Does boardroom gender diversity affect firm risk? Journal of Corporate Finance. 36, 26-53. http://doi.org/10.1016/j.jcorpfin.2015.10.003

Variable Definition SAMPLE Binary indicator that equals 1 if the firm belongs in our sample, and 0 otherwise. LAMBDA The inverse Mills ratio which controls for sample selection bias. **IRISKMM** Square root of 250 times standard deviation of the residuals from the market model regression (Data source: DataStream). ln(TRISK) Natural logarithm of one plus TRISK. ln(IRISK) Natural logarithm of one plus IRISK. SD3ROA The volatility (standard deviation) of Return on Assets in three-year overlapping periods (Data source: DataStream). The volatility (standard deviation) of Return on Assets in five-year overlapping periods SD5ROA (Data source: DataStream). SD3ROE The volatility (standard deviation) of Return on Equity in three-year overlapping periods (Data source: DataStream). SD5ROE The volatility (standard deviation) of Return on Equity in five-year overlapping periods (Data source: DataStream). Trust index is calculated by 100 + (% most people can be trusted) - (% cannot be too)SHTRUST INDEX careful) and is measured at the country of origin of the foreign shareholder. Data source: Medrano (2011). SHTRUST_INTP Represents the level of trust in the country of origin of the foreign shareholder, where the level of trust is measured by the percentage of respondents that most people can be trusted. In contrast to the SHTRUST measure in the paper we linearly interpolate the values between two adjacent surveys (Data source: WVS). GENETDIST Is the genetic distance between the population of the country of firm headquarters and the country of origin of the foreign shareholder. The measure gauges the probability that random selection by two alleles at a given locus from the population of one country and the population of the other country will be different (Data source: Enrico Spolaore's website). **OPLEV** Is the ratio of net property, plant and equipment to total assets (Data source: DataStream). DIVPAYOUT Is the ratio of common dividends divided by income before extraordinary items. Following Hasan and Habib (2019), we replace missing values of dividend to common stock with 0. (Data source: DataStream). RDEXP Research and development expenditure divided by the book value of total assets. Following Sila et al. (2016), we replace missing values of research and development expenditure with 0 (Data source: DataStream). CAPEX Capital expenditure minus sale of property divided by total assets (Data source: DataStream). STDCF Standard deviation of cash flow from operation scaled by total assets for the last three years (Data source: DataStream). HHI Herfindahl index, a measure of competition among firms in the industry (Data source: DataStream). CROSSLISTED Binary indicator that equals 1 if the firm is listed in more than one stock exchanges, and 0 otherwise (Data source: DataStream). FOR_OWN Percentage of stocks held by foreign investors (Data source: Thomson Reuters Eikon). INST_OWN Percentage of stocks held by institutional investors (Data source: Thomson Reuters Eikon). ln(GEODIST) Natural logarithm of the distance between the country of firm headquarters and the country of origin of the foreign shareholder. (Data source: the latitude and longitude data derived through Google Maps: https://developers.google.com/publicdata/docs/canonical/countries_csv). SHAREBOARDER Binary indicator that equals 1 if the country of firm headquarters and the country of origin of foreign shareholder share boarders, and 0 otherwise (Data source: CIA World Factbook) SHIDV This index explores the "degree to which people in a society are integrated into groups", measured at the country of origin of the foreign shareholder (Data source: Geert Hofstede's website).

Appendix - Variable Definitions

Variable	Definition
SHPDI	The power distance index is defined as "the extent to which the less powerful members
	of organizations and institutions (like the family) accept and expect that power is distributed unequally", measured at the country of origin of the foreign shareholder (Data source: Geert Hofstede's website).
SHUAI	The uncertainty avoidance index is defined as "a society's tolerance for ambiguity," in which people embrace or avert an event of something unexpected, unknown, or away from the status quo, measured at the country of origin of the foreign shareholder (Data source: Geert Hofstede's website).
SHMAS	In this dimension, masculinity is defined as "a preference in society for achievement, heroism, assertiveness and material rewards for success", measured at the country of origin of the foreign shareholder (Data source: Geert Hofstede's website).
IDV	This index explores the "degree to which people in a society are integrated into groups", measured at the country of firm headquarters (Data source: Geert Hofstede's website).
PDI	The power distance index is defined as "the extent to which the less powerful members of organizations and institutions (like the family) accept and expect that power is distributed unequally", measured at the country of firm headquarters (Data source: Geert Hofstede's website).
UAI	The uncertainty avoidance index is defined as "a society's tolerance for ambiguity," in which people embrace or avert an event of something unexpected, unknown, or away from the status quo, measured at the country of firm headquarters (Data source: Geert Hofstede's website).
MAS	In this dimension, masculinity is defined as "a preference in society for achievement, heroism, assertiveness and material rewards for success", measured at the country of firm headquarters (Data source: Geert Hofstede's website).
COMLAW	Binary indicator that equals 1 if a country is a common law country, and 0 otherwise (Data source: La Porta et al. (1999)).
MCAPCNT	Stock market development measured as total market capitalization divided by GDP (Data source: World Bank).
GINI	GINI index (World Bank estimate) - measures the extent to which the distribution of income (or, in some cases, consumption expenditure) among individuals or households within an economy deviates from a perfectly equal distribution. A Lorenz curve plots the cumulative percentages of total income received against the cumulative number of recipients, starting with the poorest individual or household. The Gini index measures the area between the Lorenz curve and a hypothetical line of absolute equality, expressed as a percentage of the maximum area under the line. Thus a Gini index of 0 represents perfect equality, while an index of 100 implies perfect inequality (Data source: World Bank)
SPRT	Numeric conversion of credit rating level of sovereign debt assigned by Standard and Poor's. The value 100 is assigned for AAA ratings, 95 for the next lower grade (AA) and so on until we reach the lowest grade (D) which takes the value of 5 (Data source: Trading Economics).
FITCHRT	Numeric conversion of credit rating level of sovereign debt assigned by Fitch. The value 100 is assigned for AAA ratings, 95 for the next lower grade (AA) and so on until we reach the lowest grade (D) which takes the value of 5 (Data source: Trading Economics).
MOODYSRT	Numeric conversion of credit rating level of sovereign debt assigned by Moody's. The value 100 is assigned for AAA ratings, 95 for the next lower grade (AA) and so on until we reach the lowest grade (D) which takes the value of 5 (Data source: Trading Economics).
RULAW	Rule of Law captures perceptions of the extent to which agents have confidence in and abide by the rules of society, and in particular the quality of contract enforcement, property rights, the police, and the courts, as well as the likelihood of crime and violence. Percentile rank indicates the country's rank among all countries covered by the aggregate indicator, with 0 corresponding to lowest rank, and 100 to highest rank (Data source: World Bank).

Variable	Definition
CORRUP	Control of corruption captures perceptions of the extent to which public power is
	exercised for private gain, including both petty and grand forms of corruption, as well
	as "capture" of the state by elites and private interests. Percentile rank indicates the
	country's rank among all countries covered by the aggregate indicator, with 0
	corresponding to the lowest rank, and 100 to the highest rank (Data source: World
	Bank).
VOICACC	Voice and accountability captures perceptions of the extent to which a country's citizens
	are able to participate in selecting their government, as well as freedom of expression,
	freedom of association, and a free media. Percentile rank indicates the country's rank
	among all countries covered by the aggregate indicator, with 0 corresponding to lowest
	rank, and 100 to highest rank (Data source: World Bank).
GOVEFF	Government Effectiveness captures perceptions of the quality of public services, the
	quality of the civil service and the degree of its independence from political pressures,
	the quality of policy formulation and implementation, and the credibility of the
	government's commitment to such policies. Percentile rank indicates the country's rank
	among all countries covered by the aggregate indicator, with 0 corresponding to lowest
	rank, and 100 to highest rank (Data source: World Bank).

Appendix A of the Manuscript.										
Variable	1	2	3	4	5	6	7	8	9	10
1. TRISK	1.00									
2. IRISK	0.99***	1.00								
3. SHTRUST	-0.07***	-0.07***	1.00							
4. ROA	-0.36***	-0.37***	-0.03***	1.00						
5. ln(ASSETS)	-0.56***	-0.59***	-0.06***	0.44***	1.00					
6. LEVERAGE	-0.13***	-0.14***	-0.02**	0.10***	0.24***	1.00				
7. MB	-0.09***	-0.10***	0.02*	0.25***	0.09***	0.31***	1.00			
8. ln(AGE)	-0.29***	-0.29***	0.02*	0.15***	0.38***	0.06***	0.03***	1.00		
9. BIG4	-0.33***	-0.35***	-0.03***	0.19***	0.35***	0.09***	0.04***	0.09***	1.00	
10. GDPGR	-0.05***	-0.04***	0.00	0.06***	0.03***	-0.01	0.03***	-0.07***	-0.14***	1.00
11. INVPROT	0.20***	0.21***	0.13***	-0.12***	-0.34***	-0.10***	0.00	-0.16***	0.01	-0.18***

Table IA 1 Pearson correlation matrix (N = 12,496)

This table presents the Pearson correlation matrix for the variables used in our analyses. The continuous variables are winsorized at the 1st and 99th percentiles. All variables are defined in the Appendix A of the Manuscript

Table IA 2 Foreign shareholder trust and firm risk, after controlling for sample selection bias

This table presents the 2SLS analysis (Columns 2 and 3) of the effect of foreign shareholder trust on firm market risk, augmenting our model for the inverse Mills ratio to control for sample selection bias (LAMBDA). Column 1 presents the first stage, in which the dependent variable is an indicator equal to 1 if the frim belongs in our sample (0 otherwise) and utilises the entire universe of DataStream firms with data available for our empirical model. The dependent variable in Column 2 is the standard deviation of each firm's weekly stock returns in each year, multiplied by the square root of 250 (TRISK). The dependent variable in Column 3 is the standard deviation of the residuals from the market model regression augmented with Fama-French return factors, multiplied by the square root of 250 (IRISK). The z-statistics in parentheses are based on heteroskedasticity corrected robust standard errors, clustered on firms. The continuous variables are winsorized at the 1st and 99th percentiles The statistical significance at the 10%, 5%, and 1% levels are indicated by *, **, and ***. All variables are defined in the Appendix or in the Appendix A of the Manuscript

	(1)	(1)	(2)
	First-Stage	Heckman	correction
Dependent Variables:	SAMPLE	TRISK	IRISK
SHTRUST	•	-1.960***	-1.682***
		(-3.00)	(-2.70)
ROA		-0.108***	-0.101***
		(-5.04)	(-4.87)
ln(ASSETS)	0.030***	-0.089***	-0.096***
	(9.50)	(-17.51)	(-19.76)
LEVERAGE	-0.006	0.006	0.002
	(-1.06)	(0.81)	(0.31)
MB	-0.001	-0.002	-0.002
	(-0.77)	(-1.10)	(-1.11)
ln(AGE)	0.038***	-0.056***	-0.051***
	(4.47)	(-4.20)	(-4.00)
BIG4		-0.313***	-0.322***
		(-12.75)	(-13.71)
GDPGR		0.006	0.004
		(1.20)	(0.99)
INVPROT		0.172***	0.159***
		(2.97)	(2.91)
LAMBDA		0.081*	0.080**
		(1.91)	(1.98)
(intercept)	1.049***	3.264***	3.272***
	(4.29)	(12.28)	(12.95)
Year & Industry dummies	Yes	Yes	Yes
Country dummies	Yes	No	No
Hausman test		42.613***	34.438***
Mean VIF	1.110	1.329	1.341
Observations	40,580	12,496	12,496
First stage	(1)	(1)	(2)
SHETHFR		-0.094***	-0.094***
		(-7.75)	(-7.75)
F-Statistic		59.99	59.99
Partial R ²		0.0267	0.0267
Control Variables		Yes	Yes
Year & Industry dummies	•	Yes	Yes

Table IA 3 Robustness of main results using alternative measures of risk

This table reports the 2SLS analysis results of the effect of foreign shareholder trust on alternative specifications and logarithmic transformations of the dependent variables. The dependent variable in Column 1 is the standard deviation of the residuals from the market model regression, multiplied by the square root of 250 (IRISKMM). The dependent variables in Columns 2 and 3 are the natural logarithm of one plus TRISK and IRISK, respectively. The dependent variable in Column 4 is the standard deviation of ROA in three-year overlapping periods (SD3ROA). The dependent variable in Column 5 is the standard deviation of ROA in five-year overlapping periods (SD5ROA). The dependent variable in Column 6 is the standard deviation of ROE in three-year overlapping periods (SD3ROE). The dependent variable in Column 7 is the standard deviation of ROE in five-year overlapping periods (SD5ROE). The z-statistics in parentheses are based on heteroskedasticity corrected robust standard errors, clustered on firms. The continuous variables are winsorized at the 1st and 99th percentiles The statistical significance at the 10%, 5%, and 1% levels are indicated by *, **, and ***. All variables are defined in the Appendix A of the Manuscript

	(1)	(2)	(3)	(4)	(5)	(6)	(7)
Dependent Variables:	IRISKMM	ln(TRISK)	ln(IRISK)	SD3ROA	SD5ROA	SD3ROE	SD5ROE
SHTRUST	-1.736***	-1.985***	-1.707***	-0.688***	-1.056***	-0.912***	-1.404***
	(-2.74)	(-3.03)	(-2.73)	(-2.59)	(-2.78)	(-2.64)	(-2.86)
ROA	-0.103***	-0.108***	-0.101***	-0.313***	-0.477***	-0.169***	-0.279***
	(-4.87)	(-5.02)	(-4.85)	(-17.44)	(-22.28)	(-7.79)	(-8.74)
ln(ASSETS)	-0.100***	-0.091***	-0.098***	-0.027***	-0.038***	-0.045***	-0.062***
	(-20.53)	(-18.23)	(-20.52)	(-12.58)	(-12.28)	(-17.26)	(-16.64)
LEVERAGE	0.003	0.007	0.003	-0.005	0.001	0.003	-0.013
	(0.44)	(0.91)	(0.41)	(-1.48)	(0.26)	(0.41)	(-1.21)
MB	-0.002	-0.002	-0.002	0.000	-0.002	-0.001	0.002
	(-1.13)	(-1.09)	(-1.10)	(0.03)	(-0.85)	(-0.46)	(0.47)
ln(AGE)	-0.056***	-0.060***	-0.055***	0.013**	0.020***	-0.016**	-0.034***
	(-4.45)	(-4.60)	(-4.41)	(2.57)	(2.74)	(-2.49)	(-3.36)
BIG4	-0.322***	-0.310***	-0.319***	-0.051***	-0.081***	-0.099***	-0.138***
	(-13.56)	(-12.66)	(-13.61)	(-6.18)	(-6.81)	(-7.32)	(-6.98)
GDPGR	0.006	0.007	0.006	-0.003*	-0.003	-0.007***	-0.011***
	(1.40)	(1.57)	(1.37)	(-1.96)	(-1.44)	(-2.98)	(-3.15)
INVPROT	0.180***	0.193***	0.179***	0.024	0.058**	0.094***	0.118***
	(3.26)	(3.32)	(3.30)	(1.16)	(1.97)	(3.00)	(2.59)
(intercept)	3.427***	3.347***	3.354***	0.678***	0.893***	1.180***	1.687***
	(13.29)	(12.53)	(13.20)	(7.72)	(7.64)	(10.19)	(10.11)
Year & Industry dummies	Yes						
Hausman test	35.185***	43.473***	35.24***	9.212***	12.87***	7.931***	9.488***
Mean VIF	1.327	1.313	1.327	1.372	1.426	1.231	1.242
Observations	12,496	12,496	12,496	12,258	11,437	12,024	11,212
First stage	(1)	(2)	(3)	(4)	(5)	(6)	(7)
SHETHFR	-0.094***	-0.094***	-0.094***	-0.094***	-0.098***	-0.095***	-0.099***
	(-7.73)	(-7.73)	(-7.73)	(-16.95)	(-16.72)	(-16.92)	(-16.68)
F-Statistic	59.79	59.79	59.79	287.28	279.66	286.32	278.06
Partial R ²	0.0266	0.0266	0.0266	0.0269	0.0282	0.0274	0.0288
Control Variables	Yes						
Year & Industry dummies	Yes						

Table IA 4 Debustmess of	·	altannationa maaaannaa	of fourious about old.		ale dim a aminia .	
Table IA 4 Kobusiness of	main results using	allernalive measures (of foreign snarenoide	er irusi, as weil as e	xciuding crisis i	perioa
	. mam resards asme	alternative measures	or rorongin binaronora	or trabt, as worr as e	norading erioro	pence

This table reports the 2SLS analysis results of the effect of foreign shareholder trust on firm risk a) using alternative specifications of foreign shareholder trust (SHTRUST_INDEX in Columns 1 and 2 and linearly interpolated values of shareholder trust between two adjacent surveys of WVS (SHTRUST_INTP) in Columns 3 and 4); b) using the genetic distance between the population of the country of firm headquarters and that of the origin of foreign shareholder (GENETDIST) as an alternative instrument for SHTRUST (Columns 5 and 6); and c) alternative samples (i.e., excluding the years 2015 and 2016 as they are not covered through WVS (Columns 7 and 8) and excluding the crisis period (Columns 9 and 10)). The dependent variable in Columns 1, 3, 5, 7, and 9 is the standard deviation of each firm's weekly stock returns in each year, multiplied by the square root of 250 (TRISK). The dependent variable in Columns 2, 4, 6, 8 and 10 is the standard deviation of the residuals from the market model regression augmented with Fama-French return factors, multiplied by the square root of 250 (IRISK). The z-statistics in parentheses are based on heteroskedasticity corrected robust standard errors, clustered on firms. The continuous variables are winsorized at the 1st and 99th percentiles The statistical significance at the 10%, 5%, and 1% levels are indicated by *, **, and ***. All variables are defined in the Appendix or in the Appendix A of the Manuscript.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)
	Alternat	ive trust	CUTDUCT	internolated	Using alternativ	ve instrument for	Excluding 201	5&2016 (not	Excluding the	e crisis period
	mea	sure	SHIKUSI	interpolated	SHT	RUST	covered thro	ough WVS)	(2007	-2009)
Dependent Variables:	TRISK	IRISK	TRISK	IRISK	TRISK	IRISK	TRISK	IRISK	TRISK	IRISK
SHTRUST_INDEX	-0.022***	-0.019***								
	(-2.64)	(-2.66)								
SHTRUST_INTP			-2.156***	-1.855***						
			(-2.91)	(-2.67)						
SHTRUST					-12.118**	-12.562**	-2.658***	-2.312***	-1.032***	-0.869**
					(-2.18)	(-2.21)	(-2.79)	(-2.61)	(-2.77)	(-2.42)
ROA	-0.116***	-0.108***	-0.096***	-0.091***	-0.115***	-0.108***	-0.101***	-0.095***	-0.123***	-0.115***
	(-5.08)	(-4.92)	(-4.52)	(-4.41)	(-2.74)	(-2.60)	(-4.59)	(-4.46)	(-4.84)	(-4.66)
ln(ASSETS)	-0.090***	-0.097***	-0.102***	-0.107***	-0.079***	-0.086***	-0.097***	-0.103***	-0.087***	-0.092***
	(-15.40)	(-17.84)	(-15.25)	(-16.97)	(-6.71)	(-7.17)	(-17.66)	(-19.71)	(-17.49)	(-19.27)
LEVERAGE	0.008	0.004	0.006	0.002	0.023	0.018	0.005	0.000	0.007	0.005
	(0.86)	(0.45)	(0.67)	(0.24)	(1.12)	(0.86)	(0.53)	(0.04)	(0.87)	(0.62)
MB	-0.002	-0.002	-0.002	-0.002	-0.000	0.000	-0.002	-0.002	-0.004*	-0.004*
	(-0.82)	(-0.86)	(-0.95)	(-0.98)	(-0.04)	(0.06)	(-1.05)	(-1.02)	(-1.84)	(-1.86)
ln(AGE)	-0.052***	-0.048***	-0.057***	-0.052***	-0.036	-0.030	-0.060***	-0.055***	-0.058***	-0.054***
	(-3.03)	(-3.01)	(-4.03)	(-3.92)	(-1.05)	(-0.87)	(-4.25)	(-4.07)	(-4.53)	(-4.35)
BIG4	-0.321***	-0.328***	-0.278***	-0.292***	-0.210***	-0.217***	-0.334***	-0.343***	-0.290***	-0.296***
	(-10.41)	(-11.46)	(-10.00)	(-11.07)	(-3.31)	(-3.35)	(-12.27)	(-13.26)	(-11.82)	(-12.50)
GDPGR	0.007	0.006	0.003	0.002	0.017	0.016	0.008	0.007	0.018***	0.018***
	(1.30)	(1.17)	(0.62)	(0.51)	(1.63)	(1.48)	(1.52)	(1.39)	(3.34)	(3.47)
INVPROT	0.285***	0.260***	0.061	0.066	0.342	0.361	0.221***	0.208***	0.082*	0.074
	(2.84)	(2.84)	(1.16)	(1.36)	(1.54)	(1.59)	(3.10)	(3.11)	(1.71)	(1.62)
(intercept)	3.862***	3.803***	3.773***	3.720***	5.403***	5.539***	3.606***	3.578***	2.993***	2.998***
	(8.54)	(9.08)	(10.29)	(10.76)	(4.30)	(4.32)	(10.87)	(11.41)	(12.97)	(13.44)
Year & Industry dummies	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Hausman test	44.278***	35.828***	30.495***	23.804***	112.45***	129.056***	34.661***	28.43***	34.204***	27.522***
Mean VIF	1.316	1.330	1.311	1.324	1.308	1.323	1.325	1.339	1.286	1.296
Observations	12,420	12,420	12,496	12,496	8,462	8,462	10,839	10,839	9,073	9,073

(continued on next page)

Table IA 4 (continued))									
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)
	Alternat	ive trust	SHTRUST	interpolated	Using alternativ	ve instrument for	Excluding 201	15&2016 (not	Excluding the	e crisis period
	mea	sure		r	SHT	RUST	covered thro	ough WVS)	(2007-2009)	
Dependent Variables:	TRISK	IRISK	TRISK	IRISK	TRISK	IRISK	TRISK	IRISK	TRISK	IRISK
First stage	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)
SHETHFR	-9.036***	-9.036***	-0.086***	-0.086***			-0.071***	-0.071***	-0.163***	-0.163***
	(-4.17)	(-4.17)	(-4.92)	(-4.92)			(-6.09)	(-6.09)	(-11.28)	(-11.28)
GENETDIST					-0.397**	-0.397**				
					(-2.40)	(-2.40)				
F-Statistic	17.39	17.39	24.20	24.20	15.76	15.76	37.13	37.13	127.17	127.17
Partial R ²	0.0076	0.0076	0.0119	0.0119	0.015	0.015	0.0163	0.0163	0.0693	0.0693
Control Variables	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Year & Industry dummies	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes

Table IA 5 Foreign shareholder trust and firm risk, using alternative thresholds of foreign blockholders

This table reports the 2SLS analysis results of foreign shareholder trust on firm market risk, using alternative thresholds of foreign blockholders, namely 10% (Columns 1 and 2), 20% (Columns 3 and 4) and 25% (Columns 5 and 6). The dependent variable in Columns 1, 3 and 5 is the standard deviation of each firm's weekly stock returns in each year, multiplied by the square root of 250 (TRISK). The dependent variable in Columns 2, 4 and 6 is the standard deviation of the residuals from the market model regression augmented with Fama-French return factors, multiplied by the square root of 250 (IRISK). The z-statistics in parentheses are based on heteroskedasticity corrected robust standard errors, clustered on firms. The continuous variables are winsorized at the 1st and 99th percentiles The statistical significance at the 10%, 5%, and 1% levels are indicated by *, **, and ***. All variables are defined in Appendix A of the Manuscript.

	(1)	(2)	(3)	(4)	(5)	(6)
	Foreign share	cholder $\geq 10\%$	Foreign share	cholder $\geq 20\%$	Foreign share	holder $\geq 25\%$
Dependent Variables:	TRISK	IRISK	TRISK	IRISK	TRISK	IRISK
SHTRUST	-1.583***	-1.468**	-1.529**	-1.467**	-1.583**	-1.544**
	(-2.67)	(-2.56)	(-2.48)	(-2.43)	(-2.50)	(-2.49)
ROA	-0.097***	-0.091***	-0.087***	-0.083***	-0.071***	-0.067**
	(-4.23)	(-4.09)	(-3.32)	(-3.22)	(-2.59)	(-2.51)
ln(ASSETS)	-0.093***	-0.099***	-0.091***	-0.097***	-0.088***	-0.093***
	(-15.93)	(-17.52)	(-12.48)	(-13.50)	(-10.77)	(-11.62)
LEVERAGE	0.001	-0.002	0.001	-0.001	-0.000	-0.002
	(0.15)	(-0.27)	(0.09)	(-0.14)	(-0.03)	(-0.25)
MB	-0.002	-0.002	-0.000	-0.000	-0.001	-0.001
	(-0.75)	(-0.72)	(-0.08)	(-0.06)	(-0.38)	(-0.36)
ln(AGE)	-0.076***	-0.071***	-0.087***	-0.081***	-0.098***	-0.091***
	(-4.87)	(-4.69)	(-4.16)	(-4.00)	(-4.31)	(-4.13)
BIG4	-0.327***	-0.331***	-0.365***	-0.366***	-0.371***	-0.371***
	(-11.90)	(-12.41)	(-10.56)	(-10.79)	(-9.71)	(-9.89)
GDPGR	0.008	0.006	0.004	0.001	0.002	-0.001
	(1.42)	(1.10)	(0.55)	(0.19)	(0.27)	(-0.11)
INVPROT	0.193***	0.187***	0.244***	0.240***	0.261***	0.261***
	(3.22)	(3.26)	(3.39)	(3.44)	(3.39)	(3.48)
(intercept)	3.219***	3.248***	3.147***	3.177***	3.166***	3.190***
	(12.37)	(12.86)	(10.82)	(11.14)	(10.25)	(10.49)
Year & Industry dummies	Yes	Yes	Yes	Yes	Yes	Yes
Hausman test	34.03***	31.489***	31.731***	31.128***	33.426***	33.71***
Mean VIF	1.301	1.312	1.301	1.309	1.302	1.310
Observations	9,734	9,734	6,517	6,517	5,679	5,679
First stage	(1)	(2)	(3)	(4)	(5)	(6)
SHETHFR	-0.116***	-0.116***	-0.144***	-0.144***	-0.151***	-0.151***
	(-8.39)	(-8.39)	(-8.19)	(-8.19)	(-7.82)	(-7.82)
F-Statistic	70.40	70.40	67.03	67.03	61.22	61.22
Partial R ²	0.0368	0.0368	0.0497	0.0497	0.0544	0.0544
Control Variables	Yes	Yes	Yes	Yes	Yes	Yes
Year & Industry dummies	Yes	Yes	Yes	Yes	Yes	Yes

errors, clustered on firms (Columns 1 at 5%, and 1% levels are indicated by *, **	nd 2) or on countries (Columr *, and ***, All variables are d	as 3 to 6). The continue of the continue of the second second second second second second second second second s	ous variables are wins x A of the Manuscript.	orized at the 1st and	99th percentiles The statistic	al significance at the 10%,
,	(1)	(2)	(3)	(4)	(5)	(6)
	Include co	ountry FEs	Cluster SEs a	t country level	Include country FEs and	cluster SEs at country level
Dependent Variables:	TRISK	IRISK	TRISK	IRISK	TRISK	IRISK
SHTRUST	-2.088***	-1.856***	-1.985***	-1.707***	-2.088***	-1.856***
	(-3.36)	(-3.13)	(-2.82)	(-2.72)	(-2.80)	(-2.67)
ROA	-0.054***	-0.046**	-0.108***	-0.101***	-0.054**	-0.046**
	(-2.58)	(-2.26)	(-4.08)	(-3.96)	(-2.21)	(-1.96)
ln(ASSETS)	-0.132***	-0.143***	-0.091***	-0.098***	-0.132***	-0.143***
	(-19.90)	(-22.55)	(-9.42)	(-10.65)	(-12.15)	(-14.57)
LEVERAGE	0.026***	0.024***	0.007	0.003	0.026***	0.024***
	(3.66)	(3.58)	(0.92)	(0.40)	(3.06)	(3.00)
MB	-0.003	-0.003	-0.002	-0.002	-0.003	-0.003
	(-1.51)	(-1.51)	(-1.09)	(-1.09)	(-1.50)	(-1.51)
ln(AGE)	-0.063***	-0.056***	-0.060***	-0.055***	-0.063***	-0.056***
	(-4.87)	(-4.57)	(-6.70)	(-6.31)	(-5.76)	(-5.39)
BIG4	-0.182***	-0.188***	-0.310***	-0.319***	-0.182***	-0.188***
	(-7.26)	(-7.80)	(-11.39)	(-12.40)	(-6.33)	(-6.74)
GDPGR	0.006	0.002	0.007*	0.006	0.006	0.002
	(1.32)	(0.56)	(1.76)	(1.53)	(1.29)	(0.58)
INVPROT	-0.651*	-0.686**	0.193***	0.179***	-0.651*	-0.686**
	(-1.92)	(-2.13)	(2.70)	(2.88)	(-1.88)	(-2.07)
(intercept)	4.189***	4.274***	3.347***	3.354***	4.189***	4.274***
	(12.79)	(13.99)	(12.94)	(13.96)	(9.79)	(10.77)
Year & Industry dummies	Yes	Yes	Yes	Yes	Yes	Yes
Country dummies	Yes	Yes	No	No	Yes	Yes
Hausman test	49.387***	42.793***	43.472***	35.239***	49.387***	42.793***
Mean VIF	1.313	1.327	1.313	1.327	1.313	1.327
Observations	12,496	12,496	12,496	12,496	12,496	12,496
First stage	(1)	(2)	(3)	(4)	(5)	(6)
SHETHFR	-0.100***	-0.100***	-0.094***	-0.094***	-0.100***	-0.100***
	(-7.67)	(-7.67)	(-4.91)	(-4.91)	(-5.50)	(-5.50)
F-Statistic	58.86	58.86	24.09	24.09	30.27	30.27
Partial R ²	0.029	0.029	0.0266	0.0266	0.029	0.029

This table reports the 2SLS analysis results of the effect of foreign shareholder trust on firm market risk, after a) accounting for country FEs and using heteroskedasticity corrected robust standard errors clustered on firms (Columns 1 and 2); b) using heteroskedasticity corrected robust standard errors clustered on countries (Columns 3 and 4); and c) accounting for country FEs and using heteroskedasticity corrected robust standard errors clustered on countries (Columns 1, 3 and 5); and c) accounting for country FEs and using heteroskedasticity corrected robust standard errors clustered on countries (Columns 5 and 6). The dependent variable in Columns 1, 3 and 5 is the standard deviation of each firm's weekly stock returns in each year, multiplied by the square root of 250 (TRISK). The dependent variable in Columns 2, 4 and 6 is the standard deviation of the residuals from the market model regression augmented with Fama-French return factors, multiplied by the square root of 250 (IRISK). The z-statistics in parentheses are based on heteroskedasticity corrected robust standard errors clustered robust standard errors clustered robust standard errors (IRISK).

Table IA 6 Robustness of main results, after considering for country fixed effects and clustering standard errors at country level

(continued on next page)
Table IA 6 (continued)

	(1)	(1) (2) (3)		(4)	(5)	(6)
	Include co	Include country FEs		country level	Include country FEs and c	luster SEs at country level
Dependent Variables:	TRISK	IRISK	TRISK	IRISK	TRISK	IRISK
Control Variables	Yes	Yes	Yes	Yes	Yes	Yes
Year & Industry dummies	Yes	Yes	Yes	Yes	Yes	Yes
Country dummies	Yes	Yes	No	No	Yes	Yes

of cash flow from operation parentheses are based on h at the 10%, 5%, and 1% let	of cash flow from operation scaled by total assets for the last three years. HHI is the Herfindahl index and represents a measure of competition among firms in the industry. The z-statistics in parentheses are based on heteroskedasticity corrected robust standard errors, clustered on firms. The continuous variables are winsorized at the 1st and 99th percentiles The statistical significance at the 10%, 5%, and 1% levels are indicated by *, **, and ***. All variables are defined in the Appendix or in the Appendix A of the Manuscript.											
Den en deut Venieklass	(1) TDICK	(2)	(3)	(4)	(5) TDISK	(6)	(7) TDISK	(8)	(9)	(10)	(11) TDISK	(12)
Dependent Variables:	IRISK	IRISK	IRISK	IRISK	IRISK	IRISK	IRISK	IRISK	IRISK	IKISK	IRISK	IRISK
SHTRUST	-1.906***	-1.651***	-1.969***	-1.692***	-1.996***	-1./1/***	-1.866***	-1.600***	-1.606***	-1.3/1**	-1.982***	-1.704***
	(-3.04)	(-2.77)	(-3.03)	(-2.73)	(-3.04)	(-2.75)	(-2.99)	(-2.69)	(-2.77)	(-2.48)	(-3.06)	(-2.76)
OPLEV	-0.047	-0.031										
DUDANOUT	(-0.75)	(-0.52)	0.1.4.5.4.4.4.4	0.105.000								
DIVPAYOUT			-0.146***	-0.13/***								
DDEVD			(-10.15)	(-10.18)	0 45 4**	0.202**						
RDEAP					0.454**	0.393**						
					(2.40)	(2.15)	0.00	0.057				
CAPEX							-0.026	-0.057				
(TDCE							(-0.16)	(-0.37)	0 105***	0 102***		
SIDCF									0.125^{***}	0.123^{***}		
11111									(3.78)	(3.83)	0 202***	0 274***
нні											-0.393****	-0.3/4***
DOA	0 102***	0 116***	0 109***	0 102***	0 102***	0.007***	0 124***	0 116***	0.061**	0.052**	(-0.33)	(-0.33)
KUA	-0.125^{++++}	-0.110****	-0.108****	-0.102^{+++}	-0.105^{++++}	-0.097	$-0.124^{+0.12}$	-0.110	-0.001	-0.033^{++}	-0.104^{+++}	-0.097
$\ln(\Lambda SSETS)$	(-4.87)	(-4.72)	(-3.07)	(-4.90)	(-4.//)	(-4.02)	(-3.00)	(-4.64)	(-2.17)	(-1.97)	(-4.01)	(-4.03)
III(ASSE1S)	(17.50)	(10.84)	(17.78)	(20.11)	(17.71)	(10.07)	(17.54)	(10.70)	(17.40)	(10.75)	(18.72)	-0.104
LEVEDACE	(-17.39)	(-19.64)	(-17.78)	(-20.11)	(-17.71)	(-19.97)	(-17.34)	(-19.79)	(-17.40)	(-19.73)	(-16.72)	(-20.89)
LEVERAGE	(1.47)	(0.94)	(0.57)	(0.000)	(1.06)	(0.55)	(1.65)	(1, 14)	(1.60)	(1, 12)	(1.32)	(0.84)
MR	(1.47)	0.004**	(0.37)	(0.00)	(1.00)	0.002	(1.05)	(1.14)	(1.09)	(1.12) 0.004**	(1.32)	0.04)
MB	(-2, 27)	(-2, 30)	(-0.85)	(-0.87)	(-1, 22)	(-1, 22)	(-2.10)	(-2, 22)	(-2,00)	(-2, 02)	(-1.27)	(-1, 28)
$\ln(AGE)$	-0.064***	-0.059***	-0.057***	-0.052***	-0.058***	-0.054***	-0.064***	-0.059***	-0.064***	-0.058***	-0.060***	-0.055***
III(AOL)	-0.004	-0.057	(-4.38)	(-4.19)	(-4.48)	(-4.30)	-0.004	(-4.78)	(-4.92)	-0.058	-0.000	(-4.41)
BIG4	-0.200***	-0.307***	-0 303***	-0.312***	_0 313***	-0 322***	-0 306***	-0 314***	-0 305***	_0 313***	_0 281***	_0 292***
D10-	(-11.90)	(-12.78)	(-12 59)	(-13.55)	(-12.84)	(-13.79)	(-12, 21)	(-13,11)	(-12.30)	(-13.14)	(-11.37)	(-1231)
GDPGR	0.007	0.006	0.008	0.006	0.008*	0.007	0.007	0.006	0.004	0.003	0.004	0.003
ODI OK	(1.49)	(1.32)	(1.62)	(1.43)	(1.78)	(1.56)	(1.59)	(1.47)	(0.80)	(0.64)	(0.78)	(0.58)
INVPROT	0 189***	0 175***	0 197***	0 183***	0 188***	0 175***	0 176***	0.162***	0 183***	0 169***	0 141**	0.130**
	(3.33)	(3.30)	(3.45)	(3.42)	(3.23)	(3.21)	(3.17)	(3.12)	(3.38)	(3.33)	(2.50)	(2.46)
	(====)	(2.2.0)	(==)	()	(*****)	()	(****)	(=)	(2.2.2)	(****)	(continued	on next page)

This table presents the 2SLS analysis regarding the effect of foreign shareholder trust on firm market risk, augmenting our model for additional firm-level control variables. The dependent variable in Columns 1, 3, 5, 7, 9, and 11 is the standard deviation of each firm's weekly stock returns in each year, multiplied by the square root of 250 (TRISK). The dependent variable in Columns 2, 4, 6, 8, 10, and 12 is the standard deviation of the residuals from the market model regression augmented with Fama-French return factors, multiplied by the square root of 250 (IRISK). OPLEV is the ratio of net property, plant and equipment to total assets. DIVPAYOUT is the ratio of common dividends divided by income before extraordinary items. RDEXP is research and development expenditure divided by the book value of total assets. CAPEX is capital expenditure minus sale of property divided by total assets. STDCF is the standard deviation

Table IA 7 Robustness of main results, after controlling for additional firm-level characteristics

74

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)
Dependent Variables:	TRISK	IRISK										
(intercept)	3.282***	3.291***	3.313***	3.322***	3.333***	3.342***	3.301***	3.315***	2.894***	2.875***	3.757***	3.744***
	(12.44)	(13.10)	(12.58)	(13.25)	(12.46)	(13.12)	(12.50)	(13.19)	(11.65)	(12.05)	(13.00)	(13.65)
Year & Industry dummies	Yes											
Hausman test	46.076***	37.802***	43.447***	35.167***	43.985***	35.641***	43.307***	34.968***	36.285***	29.03***	42.372***	34.219***
Mean VIF	1.275	1.288	1.296	1.308	1.297	1.310	1.276	1.288	1.462	1.475	1.309	1.321
Observations	11,756	11,756	12,496	12,496	12,496	12,496	11,678	11,678	11,505	11,505	12,496	12,496
First stage	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)
SHETHFR	-0.099***	-0.099***	-0.094***	-0.094***	-0.094***	-0.094***	-0.097***	-0.097***	-0.103***	-0.103***	-0.094***	-0.094***
	(-7.93)	(-7.93)	(-7.73)	(-7.73)	(-7.73)	(-7.73)	(-7.80)	(-7.80)	(-8.14)	(-8.14)	(-7.76)	(-7.76)
F-Statistic	62.86	62.86	59.79	59.79	59.82	59.82	60.88	60.88	66.24	66.24	60.24	60.24
Partial R ²	0.0292	0.0292	0.0266	0.0266	0.0266	0.0266	0.0287	0.0287	0.0315	0.0315	0.0267	0.0267
Control Variables	Yes											
Year & Industry dummies	Yes											

Table IA 7 (continued)

1 and 2); b) the stake owned by foreign inve	estors (FOR OW)	N – Columns 3 and	1 4); c) the stake ov	wheel by institution	al investors (INST	OWN – Column	is 5 and 6); and d)	both foreign and
institutional ownership (Columns 7 and 8). T	The dependent var	iable in Columns 1	, 3, 5 and 7 is the st	andard deviation o	of each firm's week	- ly stock returns in	each year, multip	lied by the square
root of 250 (TRISK). The dependent variable	e in Columns 2, 4	, 6 and 8 is the stan	dard deviation of th	ne residuals from the	he market model re	gression augmente	ed with Fama-Fren	hch return factors,
multiplied by the square root of 250 (IRISK). The z-statistics	in parentheses are	based on heteroske	dasticity corrected	robust standard er	rors, clustered on	firms. The continu	ous variables are
winsorized at the 1st and 99th percentiles The	e statistical signifi	cance at the 10%, 5	5%, and 1% levels a	re indicated by *, *	**, and ***. All va	riables are defined	in the Appendix o	r in the Appendix
A of the Manuscript.	-							
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Dependent Variables:	TRISK	IRISK	TRISK	IRISK	TRISK	IRISK	TRISK	IRISK
SHTRUST	-2.061***	-1.801***	-1.919***	-1.666***	-1.365**	-1.126**	-1.314**	-1.070**
	(-3.12)	(-2.86)	(-2.99)	(-2.71)	(-2.29)	(-1.97)	(-2.21)	(-2.02)
CROSSLISTED	-0.063**	-0.077***						
	(-2.34)	(-3.04)						
FOR_OWN			-0.107*	-0.067			0.313***	0.347***
			(-1.87)	(-1.22)			(4.07)	(4.71)
INST_OWN					-0.439***	-0.412***	-0.613***	-0.605***
					(-8.77)	(-8.59)	(-8.69)	(-8.97)
ROA	-0.107***	-0.101***	-0.108***	-0.101***	-0.099***	-0.093***	-0.094***	-0.088***
	(-5.01)	(-4.83)	(-5.07)	(-4.88)	(-4.69)	(-4.53)	(-4.44)	(-4.25)
ln(ASSETS)	-0.091***	-0.097***	-0.092***	-0.099***	-0.085***	-0.093***	-0.081***	-0.088***
	(-18.04)	(-20.28)	(-18.46)	(-20.64)	(-18.15)	(-20.51)	(-17.18)	(-19.42)
LEVERAGE	0.007	0.004	0.007	0.003	0.006	0.002	0.005	0.001
	(0.98)	(0.51)	(0.91)	(0.40)	(0.80)	(0.26)	(0.76)	(0.21)
MB	-0.002	-0.002	-0.002	-0.002	-0.002	-0.002	-0.001	-0.001
	(-1.09)	(-1.10)	(-1.08)	(-1.09)	(-0.82)	(-0.85)	(-0.74)	(-0.75)
ln(AGE)	-0.058***	-0.052***	-0.063***	-0.057***	-0.066***	-0.061***	-0.059***	-0.053***
	(-4.39)	(-4.15)	(-4.89)	(-4.60)	(-5.41)	(-5.15)	(-4.87)	(-4.53)
BIG4	-0.304***	-0.312***	-0.304***	-0.315***	-0.275***	-0.286***	-0.279***	-0.291***
	(-12.28)	(-13.15)	(-12.22)	(-13.21)	(-11.86)	(-12.81)	(-12.03)	(-13.03)
GDPGR	0.007	0.005	0.009*	0.007	0.009*	0.007*	0.006	0.004
	(1.36)	(1.11)	(1.78)	(1.50)	(1.94)	(1.72)	(1.25)	(0.93)
INVPROT	0.212***	0.202***	0.172***	0.166***	0.132**	0.122**	0.169***	0.163***
	(3.54)	(3.60)	(3.02)	(3.10)	(2.51)	(2.46)	(3.19)	(3.26)
(intercept)	3.343***	3.349***	3.426***	3.403***	3.382***	3.386***	3.165***	3.146***
	(12.39)	(13.00)	(12.55)	(13.11)	(13.06)	(13.68)	(12.23)	(12.72)
Year & Industry dummies	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Hausman test	45.839***	38.115***	42.204***	34.873***	24.198***	18.45***	23.12***	17.393***
Mean VIF	1.301	1.314	1.308	1.320	1.313	1.326	1.437	1.449
Observations	12,496	12,496	12,496	12,496	12,496	12,496	12,496	12,496

This table presents the 2SLS analysis regarding the effect of foreign shareholder trust on firm market risk, after augmenting the model for: a) a cross-listed indicator (CROSSLISTED - Columns

Table IA 8 Robustness of main results, after controlling for additional firm-level characteristics

(continued on next page)

Table IA 8	(continued)

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Dependent Variables:	TRISK	IRISK	TRISK	IRISK	TRISK	IRISK	TRISK	IRISK
First stage	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
SHETHFR	-0.093***	-0.093***	-0.095***	-0.095***	-0.097***	-0.097***	-0.097***	-0.097***
	(-7.65)	(-7.65)	(-7.87)	(-7.87)	(-8.06)	(-8.06)	(-8.04)	(-8.04)
F-Statistic	58.52	58.52	61.93	61.93	64.89	64.89	64.68	64.68
Partial R ²	0.0263	0.0263	0.0275	0.0275	0.0286	0.0286	0.0285	0.0285
Control Variables	Yes							
Year & Industry dummies	Yes							

Table IA 9 Robustness of main results, after considering for the geographical distance and share boarders between the firm and the foreign shareholder, as well as after controlling for cultural dimensions at the country of foreign shareholder

This table presents the 2SLS analysis regarding the effect of foreign shareholder trust on firm market risk, augmenting our model for additional country-level control variables. The dependent variable in Columns 1, 3, 5 and 7 is the standard deviation of each firm's weekly stock returns in each year, multiplied by the square root of 250 (TRISK). The dependent variable in Columns 2, 4, 6 and 8 is the standard deviation of the residuals from the market model regression augmented with Fama-French return factors, multiplied by the square root of 250 (IRISK). In(GEODIST) is the natural logarithm of the distance between the country of firm headquarters and the country of origin of the foreign shareholder. SHAREBOARDER is a binary indicator that equals 1 if the country of firm headquarters and the country of origin of the foreign shareholder. SHIDV, SHPDI, SHUAI and SHMAS are Hofstede's (2001) cultural indexes capturing individualism, power distance, uncertainty avoidance and masculinity, respectively, at the country of origin of the foreign shareholder. The z-statistics in parentheses are based on heteroskedasticity corrected robust standard errors, clustered on firms. The continuous variables are winsorized at the 1st and 99th percentiles The statistical significance at the 10%, 5%, and 1% levels are indicated by *, **, and ***. All variables are defined in the Appendix or in the Appendix A of the Manuscript.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Dependent Variables:	TRISK	IRISK	TRISK	IRISK	TRISK	IRISK	TRISK	IRISK
SHTRUST	-1.457**	-1.226**	-1.887***	-1.634***	-4.092***	-3.752***	-1.743***	-1.597**
	(-2.43)	(-2.13)	(-3.11)	(-2.84)	(-2.96)	(-2.84)	(-2.59)	(-2.44)
ln(GEODIST)	0.116***	0.106***						
	(8.83)	(8.37)						
SHAREBOARDER			0.025	0.019				
			(0.60)	(0.47)				
SHIDV					-0.008**	-0.007**	-0.003*	-0.003*
					(-2.30)	(-2.32)	(-1.77)	(-1.87)
SHPDI					-0.021**	-0.020**	-0.009**	-0.009*
					(-2.54)	(-2.46)	(-2.03)	(-1.95)
SHUAI							-0.004***	-0.004***
							(-5.18)	(-4.90)
SHMAS							-0.001	-0.001
							(-0.92)	(-0.90)
ROA	-0.100***	-0.094***	-0.107***	-0.101***	-0.096***	-0.090***	-0.096***	-0.090***
	(-4.68)	(-4.53)	(-5.01)	(-4.85)	(-4.37)	(-4.24)	(-4.50)	(-4.36)
ln(ASSETS)	-0.098***	-0.105***	-0.091***	-0.098***	-0.101***	-0.107***	-0.097***	-0.104***
	(-19.41)	(-21.42)	(-17.77)	(-20.00)	(-16.36)	(-18.18)	(-19.46)	(-21.41)
LEVERAGE	0.008	0.004	0.007	0.003	0.007	0.002	0.008	0.004
	(1.11)	(0.58)	(0.92)	(0.41)	(0.82)	(0.29)	(1.18)	(0.54)
MB	-0.003	-0.003	-0.002	-0.002	-0.002	-0.002	-0.003	-0.003
	(-1.34)	(-1.34)	(-1.10)	(-1.11)	(-1.16)	(-1.13)	(-1.57)	(-1.50)
ln(AGE)	-0.067***	-0.062***	-0.060***	-0.055***	-0.086***	-0.078***	-0.076***	-0.068***
	(-5.44)	(-5.17)	(-4.67)	(-4.46)	(-5.50)	(-5.19)	(-6.08)	(-5.63)
BIG4	-0.294***	-0.304***	-0.311***	-0.320***	-0.332***	-0.339***	-0.326***	-0.333***
	(-12.86)	(-13.81)	(-12.76)	(-13.69)	(-11.39)	(-12.14)	(-13.61)	(-14.32)
GDPGR	0.003	0.003	0.008	0.006	0.002	0.001	0.001	0.001
	(0.79)	(0.61)	(1.62)	(1.41)	(0.35)	(0.21)	(0.30)	(0.15)

(continued on next page)

T 11 TAO	/ /' 1\
I anie I A Y I	continued)
10010111	(continucu)

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Dependent Variables:	TRISK	IRISK	TRISK	IRISK	TRISK	IRISK	TRISK	IRISK
INVPROT	0.123**	0.115**	0.191***	0.178***	0.195**	0.186**	0.054	0.058
	(2.38)	(2.37)	(3.37)	(3.34)	(2.40)	(2.40)	(0.98)	(1.08)
(intercept)	2.514***	2.594***	3.305***	3.323***	5.803***	5.659***	4.639***	4.592***
	(10.90)	(11.77)	(12.67)	(13.36)	(5.50)	(5.62)	(7.21)	(7.36)
Year & Industry dummies	Yes							
Hausman test	27.014***	21.174***	42.901***	35.226***	52.451***	47.433***	35.913***	32.923***
Mean VIF	1.305	1.317	1.305	1.317	1.839	1.851	1.811	1.821
Observations	12,496	12,496	12,496	12,496	11,907	11,907	11,907	11,907
First stage	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
SHETHFR	-0.099***	-0.099***	-0.100***	-0.100***	-0.060***	-0.060***	-0.107***	-0.107***
	(-8.12)	(-8.12)	(-8.00)	(-8.00)	(-5.21)	(-5.21)	(-8.91)	(-8.91)
F-Statistic	66.00	66.00	64.08	64.08	27.17	27.17	79.32	79.32
Partial R ²	0.03	0.03	0.0286	0.0286	0.0129	0.0129	0.0429	0.0429
Control Variables	Yes							
Year & Industry dummies	Yes							

Tuble in the Robustiess of main results, after controlling for additional country for or entrolling of composition near quarters
This table presents the 2SLS analysis regarding the effect of foreign shareholder trust on firm market risk, augmenting our model for additional country-level control variables. The dependent
variable in Columns 1, 3, 5, 7, 9, 11 and 14 is the standard deviation of each firm's weekly stock returns in each year, multiplied by the square root of 250 (TRISK). The dependent variable in
Columns 2, 4, 6, 8, 10, 12 and 14 is the standard deviation of the residuals from the market model regression augmented with Fama-French return factors, multiplied by the square root of 250
(IRISK). IDV, PDI, UAI and MAS are four Hofstede's (2001) culture indexes capturing individualism, power distance, uncertainty avoidance, and masculinity, respectively. MCAPCNT is total
market capitalization divided by GDP. COMLAW is a binary indicator that equals 1 for common law country. GINI index measures the extent to which the distribution of income among individuals
or households within an economy deviates from a perfectly equal distribution. SPRT, FITCHRT, and MOODYSRT represent the credit rating level of sovereign debt assigned by Standard and
Poor's, Fitch, and Moody's, respectively. The z-statistics in parentheses are based on heteroskedasticity corrected robust standard errors, clustered on firms. The continuous variables are winsorized
at the 1st and 99th percentiles The statistical significance at the 10%, 5%, and 1% levels are indicated by *, **, and ***. All variables are defined in the Appendix or in the Appendix A of the
Manuscript.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)	(13)	(14)
Dependent Variables:	TRISK	IRISK	TRISK	IRISK										
SHTRUST	-1.955***	-1.681***	-1.841***	-1.548**	-1.921***	-1.647***	-2.254***	-1.913**	-1.932***	-1.673***	-2.357***	-2.063***	-1.975***	-1.700***
	(-3.12)	(-2.81)	(-2.83)	(-2.50)	(-3.00)	(-2.70)	(-2.61)	(-2.34)	(-2.97)	(-2.69)	(-3.44)	(-3.16)	(-3.04)	(-2.74)
IDV	-0.003***	-0.004***												
	(-3.11)	(-4.14)												
PDI	-0.004***	-0.004***												
	(-3.46)	(-3.50)												
UAI	-0.001	-0.001												
	(-1.28)	(-0.70)												
MAS	0.005***	0.004***												
	(6.40)	(6.11)												
COMLAW			-0.093***	-0.104***										
			(-2.58)	(-3.03)										
MCAP_CNT					-0.001**	-0.001**								
					(-2.52)	(-2.53)								
GINI							0.011***	0.011***						
							(4.10)	(4.70)						
SPRT									0.003***	0.002*				
									(2.66)	(1.82)				
FITCHRT											0.004***	0.003***		
											(4.01)	(3.24)		
MOODYSRT													0.005***	0.004***
													(4.91)	(4.18)
ROA	-0.095***	-0.087***	-0.105***	-0.099***	-0.108***	-0.101***	-0.097***	-0.091***	-0.108***	-0.101***	-0.104***	-0.097***	-0.107***	-0.101***
	(-4.48)	(-4.24)	(-4.93)	(-4.75)	(-5.03)	(-4.86)	(-4.46)	(-4.31)	(-5.06)	(-4.88)	(-4.84)	(-4.67)	(-5.04)	(-4.87)
ln(ASSETS)	-0.105***	-0.114***	-0.094***	-0.101***	-0.092***	-0.099***	-0.091***	-0.097***	-0.090***	-0.097***	-0.094***	-0.101***	-0.089***	-0.097***
	(-17.16)	(-19.55)	(-17.86)	(-20.17)	(-18.37)	(-20.64)	(-17.28)	(-19.49)	(-18.04)	(-20.40)	(-18.08)	(-20.37)	(-18.12)	(-20.48)
LEVERAGE	0.007	0.004	0.007	0.003	0.006	0.002	0.006	0.002	0.005	0.002	0.002	-0.002	0.004	0.001
	(0.93)	(0.61)	(0.99)	(0.50)	(0.85)	(0.34)	(0.79)	(0.31)	(0.72)	(0.27)	(0.21)	(-0.22)	(0.57)	(0.11)
												(C	ontinued on	next page)

Table IA 10 Robustness of main results, after controlling for additional country-level characteristics at the country of corporate headquarters

80

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)	(13)	(14)
Dependent Variables:	TRISK	IRISK												
MB	-0.002	-0.002	-0.002	-0.002	-0.002	-0.002	-0.003	-0.003	-0.002	-0.002	-0.002	-0.002	-0.002	-0.002
	(-1.07)	(-1.06)	(-0.99)	(-0.99)	(-1.04)	(-1.05)	(-1.23)	(-1.27)	(-0.99)	(-1.03)	(-0.84)	(-0.89)	(-0.91)	(-0.95)
ln(AGE)	-0.064***	-0.060***	-0.061***	-0.056***	-0.060***	-0.055***	-0.066***	-0.060***	-0.058***	-0.054***	-0.052***	-0.048***	-0.056***	-0.052***
	(-4.91)	(-4.77)	(-4.70)	(-4.52)	(-4.64)	(-4.44)	(-4.63)	(-4.42)	(-4.46)	(-4.31)	(-3.83)	(-3.72)	(-4.27)	(-4.12)
BIG4	-0.296***	-0.306***	-0.319***	-0.329***	-0.309***	-0.318***	-0.326***	-0.334***	-0.321***	-0.326***	-0.316***	-0.322***	-0.326***	-0.332***
	(-11.93)	(-12.85)	(-13.04)	(-14.07)	(-12.69)	(-13.64)	(-12.65)	(-13.61)	(-12.85)	(-13.62)	(-12.21)	(-13.05)	(-13.05)	(-13.87)
GDPGR	0.007	0.004	0.010**	0.009*	0.009*	0.008*	0.005	0.003	0.011**	0.009*	0.013**	0.010**	0.016***	0.013***
	(1.27)	(0.89)	(1.99)	(1.90)	(1.86)	(1.67)	(0.90)	(0.57)	(2.18)	(1.77)	(2.49)	(2.07)	(2.99)	(2.59)
INVPROT	0.044	0.072	0.277***	0.272***	0.279***	0.260***	0.309***	0.285***	0.100	0.118*	0.032	0.050	0.032	0.048
	(0.75)	(1.29)	(4.54)	(4.82)	(4.02)	(3.98)	(3.68)	(3.61)	(1.48)	(1.85)	(0.46)	(0.78)	(0.48)	(0.77)
(intercept)	3.916***	3.941***	3.260***	3.257***	3.295***	3.305***	2.899***	2.864***	3.204***	3.260***	3.325***	3.373***	3.096***	3.149***
	(12.43)	(13.09)	(12.26)	(12.88)	(12.61)	(13.30)	(9.47)	(9.90)	(12.00)	(12.80)	(12.02)	(12.84)	(11.82)	(12.62)
Year & Industry dummies	Yes													
Hausman test	43.21***	35.257***	37.596***	29.461***	42.017***	33.874***	30.751***	24.083***	41.593***	34.079***	55.481***	46.004***	43.179***	34.982***
Mean VIF	1.825	1.837	1.460	1.473	1.406	1.418	1.329	1.342	1.550	1.563	1.547	1.560	1.523	1.536
Observations	12,246	12,246	12,496	12,496	12,496	12,496	11,610	11,610	12,496	12,496	12,374	12,374	12,496	12,496
First stage	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)	(13)	(14)
SHETHFR	-0.099***	-0.099***	-0.093***	-0.093***	-0.095***	-0.095***	-0.075***	-0.075***	-0.094***	-0.094***	-0.093***	-0.093***	-0.094***	-0.094***
	(-8.10)	(-8.10)	(-7.60)	(-7.60)	(-7.85)	(-7.85)	(-6.23)	(-6.23)	(-7.72)	(-7.72)	(-7.62)	(-7.62)	(-7.73)	(-7.73)
F-Statistic	65.65	65.65	57.75	57.75	61.59	61.59	38.77	38.77	59.55	59.55	58.05	58.05	59.81	59.81
Partial R ²	0.0287	0.0287	0.0259	0.0259	0.0275	0.0275	0.0194	0.0194	0.0266	0.0266	0.0256	0.0256	0.0266	0.0266
Control Variables	Yes													
Year & Industry dummies	Yes													

Table IA 10 (continued)

parentheses are based on heteroskedasticity corrected robust standard errors, clustered on firms. The continuous variables are winsorized at the 1st and 99th percentiles The statistical significance												
at the 10%, 5%, and 1% levels are indicated by *, **, and ***. All variables are defined in the Appendix A of the Manuscript.												
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)
Countries excluded:	USA		Japan		India		Canada		Australia		All five countries	
Dependent Variables:	TRISK	IRISK	TRISK	IRISK								
SHTRUST	-3.000***	-2.571***	-1.601***	-1.368**	-1.740***	-1.457**	-1.364**	-1.132**	-2.364***	-2.010***	-1.583***	-1.226**
	(-4.33)	(-4.01)	(-2.63)	(-2.34)	(-2.63)	(-2.31)	(-2.50)	(-2.17)	(-2.92)	(-2.62)	(-3.09)	(-2.55)
ROA	-0.123***	-0.115***	-0.094***	-0.088***	-0.103***	-0.097***	-0.127***	-0.119***	-0.113***	-0.106***	-0.150***	-0.141***
	(-3.82)	(-3.70)	(-4.38)	(-4.23)	(-4.84)	(-4.68)	(-5.33)	(-5.13)	(-5.10)	(-4.98)	(-3.04)	(-2.94)
ln(ASSETS)	-0.072***	-0.080***	-0.104***	-0.110***	-0.091***	-0.097***	-0.084***	-0.091***	-0.090***	-0.096***	-0.063***	-0.069***
	(-14.34)	(-16.92)	(-17.41)	(-19.09)	(-18.04)	(-20.21)	(-16.51)	(-18.71)	(-16.48)	(-18.61)	(-9.93)	(-11.43)
LEVERAGE	0.006	0.003	0.004	0.001	0.006	0.002	0.010	0.006	0.007	0.003	0.008	0.006
	(0.76)	(0.34)	(0.55)	(0.09)	(0.70)	(0.32)	(1.40)	(0.89)	(0.89)	(0.39)	(0.82)	(0.63)
MB	-0.001	-0.001	-0.002	-0.002	-0.002	-0.002	-0.003	-0.003	-0.001	-0.001	0.001	0.001
	(-0.33)	(-0.34)	(-1.03)	(-1.04)	(-0.81)	(-0.85)	(-1.46)	(-1.51)	(-0.58)	(-0.60)	(0.42)	(0.29)
ln(AGE)	-0.038***	-0.035***	-0.071***	-0.066***	-0.055***	-0.049***	-0.071***	-0.065***	-0.058***	-0.053***	-0.048***	-0.044***
	(-2.73)	(-2.74)	(-5.26)	(-5.11)	(-4.20)	(-3.91)	(-5.16)	(-4.94)	(-4.01)	(-3.89)	(-3.11)	(-2.99)
BIG4	-0.228***	-0.240***	-0.314***	-0.323***	-0.341***	-0.355***	-0.296***	-0.304***	-0.321***	-0.329***	-0.225***	-0.245***
	(-9.08)	(-10.21)	(-11.80)	(-12.62)	(-12.44)	(-13.46)	(-12.36)	(-13.23)	(-11.95)	(-12.93)	(-7.19)	(-8.21)
GDPGR	0.026***	0.024***	0.014***	0.013**	0.012**	0.013**	0.008*	0.007	0.009*	0.007	0.046***	0.045***
	(5.27)	(5.09)	(2.69)	(2.55)	(2.13)	(2.38)	(1.80)	(1.62)	(1.80)	(1.54)	(7.44)	(7.60)
INVPROT	0.184***	0.171***	0.137**	0.129**	0.165***	0.152***	0.154***	0.142***	0.261***	0.237***	-0.019	-0.019
	(2.93)	(2.97)	(2.46)	(2.46)	(2.81)	(2.76)	(2.99)	(2.94)	(3.62)	(3.54)	(-0.35)	(-0.39)
(intercept)	2.973***	2.972***	3.498***	3.495***	3.286***	3.273***	3.125***	3.148***	3.301***	3.296***	2.549***	2.516***
	(12.83)	(13.86)	(13.34)	(13.94)	(12.26)	(12.83)	(12.31)	(12.95)	(10.49)	(11.10)	(11.80)	(12.64)
Year & Industry dummies	Yes	Yes										
Hausman test	82.851***	65.021***	32.044***	25.879***	32.871***	25.639***	30.292***	23.388***	43.155***	34.223***	30.316***	19.814***
Mean VIF	1.256	1.269	1.323	1.335	1.313	1.326	1.288	1.301	1.309	1.322	1.193	1.202
Observations	11,172	11,172	11,062	11,062	11,336	11,336	11,408	11,408	11,311	11,311	6,305	6,305
First stage	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)
SHETHFR	-0.095***	-0.095***	-0.105***	-0.105***	-0.098***	-0.098***	-0.110***	-0.110***	-0.082***	-0.082***	-0.137***	-0.137***
	(-6.92)	(-6.92)	(-8.25)	(-8.25)	(-7.54)	(-7.54)	(-8.56)	(-8.56)	(-6.65)	(-6.65)	(-7.33)	(-7.33)
F-Statistic	47.86	47.86	68.06	68.06	56.91	56.91	73.32	73.32	44.25	44.25	53.80	53.80
Partial R ²	0.0265	0.0265	0.0316	0.0316	0.0279	0.0279	0.0349	0.0349	0.0213	0.0213	0.0462	0.0462
Control Variables	Yes	Yes										
Year & Industry dummies	Yes	Yes										

This table presents the 2SLS analysis of the effect of foreign shareholder trust on firm market risk after excluding firms from: a) US (Columns 1 and 2), b) Japan (Columns 3 and 4), c) India (Columns 5 and 6), d) Canada (Columns 7 and 8), e) Australia (Columns 9 and 10), and f) all five aforementioned countries (Columns 11 and 12). The dependent variable in Columns 1, 3, 5, 7, 9, and 11 is the standard deviation of each firm's weekly stock returns in each year, multiplied by the square root of 250 (TRISK). The dependent variable in Columns 2, 4, 6, 8, 10, and 12 is the standard deviation of the residuals from the market model regression augmented with Fama-French return factors, multiplied by the square root of 250 (IRISK). The z-statistics in

Table IA 11 Foreign shareholder trust and firm risk, after excluding countries with high representation

2, 4, 6, and 8 is the standard deviation of the residuals from the market model regression augmented with Fama-French return factors, multiplied by the square root of 250 (IRISK). The z- statistics in parentheses are based on heteroskedasticity corrected robust standard errors, clustered on firms. The continuous variables are winsorized at the 1st and 99th percentiles The statistical significance at the 10%, 5%, and 1% levels are indicated by *, **, and ***. All variables are defined in the Appendix or in the Appendix A of the Manuscript.										
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)		
	Low Investor Protection (RULAW)		High Investor Protection (RULAW)		Low Investor Protection (CORRUP)		High Ir (vestor Protection CORRUP)		
Dependent Variables:	TRISK	IRISK	TRISK	IRISK	TRISK	IRISK	TRISK	IRISK		
SHTRUST	-5.712**	-5.245**	-0.782	-0.712	-3.909***	-3.502***	-1.259*	-1.220*		
	(-2.55)	(-2.50)	(-1.41)	(-1.32)	(-2.88)	(-2.77)	(-1.91)	(-1.91)		
ROA	-0.154***	-0.147***	-0.073***	-0.066***	-0.139***	-0.130***	-0.074***	-0.068***		
ln(ASSETS)	(-3.84) -0.076***	(-3.77) -0.082***	(-3.05) -0.123***	(-2.85) -0.131***	(-3.11) -0.066***	(-2.99) -0.071***	(-3.21) -0.125***	(-3.01) -0.134***		
	(-7.39)	(-8.40)	(-18.66)	(-20.58)	(-7.93)	(-9.07)	(-18.30)	(-20.09)		
LEVERAGE	0.011	0.006	-0.002	-0.003	0.007	0.002	0.006	0.004		
	(0.69)	(0.39)	(-0.16)	(-0.32)	(0.52)	(0.15)	(0.60)	(0.39)		
MB	0.000	-0.000	-0.002	-0.002	-0.003	-0.003	-0.001	-0.001		
	(0.05)	(-0.06)	(-0.74)	(-0.71)	(-0.93)	(-1.07)	(-0.41)	(-0.40)		
ln(AGE)	-0.012	-0.012	-0.057***	-0.052***	-0.022	-0.023	-0.059***	-0.053***		
	(-0.23)	(-0.25)	(-4.33)	(-4.06)	(-0.65)	(-0.71)	(-4.25)	(-3.94)		
BIG4	-0.136***	-0.146***	-0.354***	-0.361***	-0.149***	-0.155***	-0.344***	-0.353***		
	(-2.93)	(-3.38)	(-9.55)	(-9.98)	(-4.46)	(-4.98)	(-9.09)	(-9.52)		
GDPGR	0.040***	0.035***	-0.016*	-0.012	0.047***	0.042***	-0.022**	-0.018**		
	(3.34)	(3.10)	(-1.94)	(-1.43)	(5.02)	(4.75)	(-2.47)	(-2.13)		
(intercept)	4.022***	3.976***	3.902***	3.928***	3.288***	3.246***	4.116***	4.165***		
	(6.11)	(6.45)	(12.50)	(12.91)	(7.61)	(8.11)	(11.68)	(12.03)		
Year & Industry dummies	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes		
Hausman test	51.271***	46.377***	11.475***	10.712***	36.676***	31.606***	19.588***	19.704***		
Mean VIF	1.260	1.270	1.358	1.375	1.230	1.240	1.384	1.401		
Observations	6,292	6,292	6,204	6,204	5,991	5,991	6,505	6,505		
First stage	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)		
SHETHFR	-0.052***	-0.052***	-0.145***	-0.145***	-0.066***	-0.066***	-0.127***	-0.127***		
	(-3.16)	(-3.16)	(-8.26)	(-8.26)	(-4.04)	(-4.04)	(-7.49)	(-7.49)		
F-Statistic	19.98	19.98	68.28	68.28	16.36	16.36	56.06	56.06		
Partial R ²	0.0083	0.0083	0.0594	0.0594	0.0138	0.0138	0.0451	0.0451		
Control Variables	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes		
Year & Industry dummies	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes		

This table presents the 2SLS analysis of the effect of foreign shareholder trust on firm market risk, after separating the sample into countries with low (high) investor protection in Columns 1 and 2 (3 and 4) when using the sample median of RULAW; and in low (high) investor protection in Columns 5 and 6 (7 and 8) when using the sample median of CORRUP. The dependent variable in Columns 1, 3, 5, and 7 is the standard deviation of each firm's weekly stock returns in each year, multiplied by the square root of 250 (TRISK). The dependent variable in Columns

Table IA 12 Variations in investor protection, foreign shareholder trust and firm risk, using alternative definitions of investor protection

Table IA 13 Variations in institutional context strength, foreign shareholder trust and firm risk, using alternative definitions of institutional context strength

This table presents the 2SLS analysis of the effect of foreign shareholder trust on firm market risk, after separating the sample into countries with weak (strong) institutional contexts in Columns 1 and 2 (3 and 4) when using the sample median of VOICACC; and in weak (strong) institutional contexts in Columns 5 and 6 (7 and 8) when using the sample median of GOVEFF. The dependent variable in Columns 1, 3, 5, and 7 is the standard deviation of each firm's weekly stock returns in each year, multiplied by the square root of 250 (TRISK). The dependent variable in Columns 2, 4, 6, and 8 is the standard deviation of the residuals from the market model regression augmented with Fama-French return factors, multiplied by the square root of 250 (IRISK). The z-statistics in parentheses are based on heteroskedasticity corrected robust standard errors, clustered on firms. The continuous variables are winsorized at the 1st and 99th percentiles The statistical significance at the 10%, 5%, and 1% levels are indicated by *, **, and ***. All variables are defined in the Appendix or in the Appendix A of the Manuscript.

	(1) (2)		(3)	(4)	(5)	(6)	(7) (8)		
	Weak Institutional		Strong Ins	stitutional	Weak Ins	stitutional	Strong Institutional		
	Context (VOICACC)		Con	text	Cor	itext	Context		
			(VOICACC)		(GOV	/EFF)	(GOVEFF)		
Dependent Variables:	TRISK	IRISK	TRISK	IRISK	TRISK	IRISK	TRISK	IRISK	
SHTRUST	- 3.042***	- 2.788***	-1.069	-0.907	-4.890**	-4.363**	-0.857	-0.805	
	(-3.04)	(-2.94)	(-1.56)	(-1.37)	(-2.52)	(-2.44)	(-1.54)	(-1.48)	
ROA	- 0.117***	- 0.111***	- 0.084***	0.076** *	- 0.144***	- 0.135***	- 0.070***	- 0.064***	
	(-3.49)	(-3.39)	(-3.32)	(-3.14)	(-3.42)	(-3.31)	(-2.97)	(-2.80)	
ln(ASSETS)	- 0.083***	- 0.089***	- 0.112***	0.120** *	- 0.069***	- 0.075***	- 0.124***	- 0.131***	
	(-10.63)	(-11.97)	(-16.97)	(-18.91)	(-7.81)	(-9.00)	(-18.64)	(-20.43)	
LEVERAGE	0.019*	0.013	0.000	-0.001	0.007	0.002	0.002	0.001	
	(1.67)	(1.24)	(0.02)	(-0.12)	(0.50)	(0.19)	(0.23)	(0.06)	
MB	-0.003	-0.003	-0.001	-0.001	-0.002	-0.002	-0.001	-0.001	
	(-1.03)	(-1.13)	(-0.36)	(-0.36)	(-0.54)	(-0.70)	(-0.36)	(-0.35)	
ln(AGE)	-0.040	-0.037	- 0.062***	- 0.056** *	-0.008	-0.009	- 0.066***	- 0.061***	
	(-1.48)	(-1.45)	(-4.62)	(-4.31)	(-0.21)	(-0.26)	(-4.82)	(-4.50)	
BIG4	- 0.230***	- 0.234***	- 0.323***	0.333** *	- 0.148***	- 0.156***	- 0.348***	- 0.359***	
	(-7.38)	(-7.89)	(-8.87)	(-9.41)	(-3.75)	(-4.30)	(-9.65)	(-10.16)	
GDPGR	0.029***	0.024***	-0.023**	-0.014	0.054***	0.048^{***}	-0.019**	-0.015*	
	(4.17)	(3.70)	(-2.55)	(-1.64)	(4.48)	(4.31)	(-2.24)	(-1.85)	
INVPROT	0.198**	0.186**	0.259***	0.265** *	0.472**	0.428**	-0.077	-0.056	
	(2.18)	(2.18)	(3.30)	(3.53) 3 340**	(2.19)	(2.16)	(-1.04)	(-0.79)	
(intercept)	3.314***	3.335***	3.372***	*	2.949***	2.939***	4.073***	4.069***	
Year & Industry	(10.00)	(10.65)	(10.36)	(10.65)	(7.68)	(8.39)	(11.73)	(11.96)	
dummies	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	
Hausman test	37.251** *	33.298** *	11.518** *	9.719** *	38.779** *	33.157** *	13.612** *	13.125** *	
Mean VIF	1.271	1.281	1.344	1.358	1.230	1.239	1.390	1.405	
Observations	6,114	6,114	6,382	6,382	6,035	6,035	6,461	6,461	
First stage	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	
	_	_	_	-	_	_	_	_	
SHETHFR	0.084***	0.084***	0.115***	0.115** *	0.053***	0.053***	0.143***	0.143***	
	(-5.15)	(-5.15)	(-6.91)	(-6.91)	(-3.20)	(-3.20)	(-8.72)	(-8.72)	
F-Statistic	20.55	20.55	47.75	47.75	10.25	10.25	/0.10	/0.10	
Partial K ²	0.0228 V	0.0228 Var	0.0369 V	0.0369 Var	0.0091 No.	0.0091 No.	0.0589 Note	0.0589 V	
Voor & Industry	r es	r es	res	r es	res	res	res	res	
dummies	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	